## DEPARTMENT OF HEALTH AND HUMAN SERVICES

## Administration for Children and Families

[Program Announcement No. ACYF/HS-2001-07]

Fiscal Year 2001 Discretionary Announcement for Head Start Family Worker Training and Credentialing Initiative; Availability of Funds and Request for Applications

**AGENCY:** Administration on Children, Youth and Families (ACYF), ACF, DHHS.

**ACTION:** Notice.

**SUMMARY:** The Administration for Children and Families (ACF), Administration on Children, Youth and Families (ACYF) announces the availability of \$1,000,000; up to \$100,000 per project for one year to support up to ten entities to design and/ or adapt competency-based training programs and curricula suitable for the training and credentialing of Head Start Family Worker Staff. Academic institutions, other training providers, and public or private non-profit or forprofit organizations are eligible to apply for projects, which will be funded on a competitive basis.

Applicants must provide assurances that if they receive funds under the announcement, the model training program required as part of the final report described in the section of this announcement entitled: Expectations and Requirements for Family Worker Training and Credentialing Projects will be established as part of the grantee's regular curricular offerings no later than one year from the date of submission of the report.

**DATES:** The closing date for receipt of applications is 5:00 P.M. EDT, July 2, 2001

ADDRESSES: Mail applications to: Head Start Family Worker Training and Credentialing Initiative, ACYF Operations Center, 1815 North Fort Myer Drive, Suite 300, Arlington, Virginia 22209.

Hand delivered courier or overnight delivery applications are accepted during the normal working hours of 8:00 a.m. to 5:00 p.m., Monday through Friday, on or prior to the established closing date.

All packages should be clearly labeled as follows:

Application for Head Start Family Worker Training and Credentialing Initiative FOR FURTHER INFORMATION CONTACT: The Head Start Discretionary Grant Support Team (1–800–351–2293) is available to answer questions concerning application requirements and to refer you to the appropriate contact person in ACYF for programmatic questions. You may e-mail your questions to: hs@lcgnet.com.

In order to determine the number of expert reviewers that will be necessary, if you plan to submit an application, you are requested to send a post card or call with the following information: the name, address, telephone and fax numbers, and e-mail address of the project director and the name of the applicant at least four weeks prior to the submission deadline date to: Head Start Family Worker Training and Credentialing Initiative, ACYF Operations Center, 1815 North Fort Myer Drive, Suite 300, Arlington, Virginia 22209.

Fiscal Year 2001 Discretionary Announcement for Head Start Family Worker Training and Credentialing Initiative

#### **Table of Contents**

application process.

This program announcement is divided into five sections:

Part I contains general information and an introductory section that describes the background of various Head Start initiatives supporting professional development, the target audience of this initiative, and the Head Start Bureau's expectations regarding this initiative and next steps.

Part II contains key program information such as a description of competitive categories, eligible applicants, project periods and applicable Head Start regulations.

Part III contains the requirements for information that must be included in each application.

Part IV contains the criteria upon which applications will be reviewed and evaluated. Part V contains a discussion of the

Appendix A includes the relevant application forms, certifications, disclosures and assurances necessary for completing and submitting the application.

Appendix B contains a listing of Competency Goals and Indicators for Head Start Staff Working with Families.

Appendix C includes a listing of the Head Start Quality Improvement Centers. The Head Start Quality Improvement Centers and Disability Services Quality Improvement Centers form a regionally-based system, composed of institutions and organizations whose common purpose is to support the continuous improvement of all grantees and delegate agencies as they work to provide high quality and effective services to children and families and address the emerging priorities of child care partnerships, Head Start expansion and welfare reform. The Training and Technical Assistance reflects a national commitment to quality

improvement, local capacity-building and ongoing evaluation.

#### Part I. Purpose and Background

#### A. Purpose

The purpose of this announcement is to solicit applications for grants for the design and/or adaptation of competency-based training programs appropriate for utilization in a national Head Start Family Worker Training and Credentialing Initiative. Grants will be awarded to develop methodologies and approaches to enhance the skills, knowledge, and effectiveness of Family Services staff who are working with parents and young children in Head Start and Early Head Start, and other early childhood and child care family support programs.

Organizations funded under this Announcement will work cooperatively with the Head Start Bureau, national experts, and national organizations in furthering this initiative.

Successful applicants will be expected to work collaboratively with local Head Start programs as well as with other service agencies and organizations involved in endeavors, which grant credit, degrees, and credentialing of Family Workers.

#### B. Background

#### **Head Start**

Head Start and Early Head Start are comprehensive child development programs which serve children from birth to age five, pregnant women, and their families. The Early Head Start program provides services to children zero to three and serves approximately 50,000 children. Head Start, which provides services to children age three to five, currently serves over 850,000 low-income families and their children through a nationwide network of approximately 2,100 grantee and delegate agencies. These agencies serve children and families through a variety of program options and service strategies including center-based, homebased, and family child care partnerships.

**Note:** In the balance of this document, the term "Head Start" refers to both Head Start and Early Head Start programs and staff, unless otherwise indicated.

Since its inception in 1965, Head Start has had a strong commitment and impressive success in involving, educating and supporting parents and families as an integral part of every local program. For example, recent research in a nationally representative sample of programs documented high levels of parent involvement and satisfaction: approximately 80% of all parents participate in home visits, parentteacher conferences, classroom observations and volunteering and over 85% of parents were very satisfied with the quality of services their child received. These findings corroborate the 1999 report of the American Customers Satisfaction Index, in which Head Start received the highest rating of any government program.

Building on this strong record of success and commitment, the initiative described in this announcement is intended to continue to strengthen the quality of services to and depth of partnerships with families by enhancing the training and effectiveness of Family Workers in all Head Start programs. It complements a broad series of related efforts to improve Head Start program quality, staff credentials, and accountability, including more specific performance standards and measures in children's literacy and language development, a new focus on child outcomes in program monitoring and self-assessment, expanded funding to upgrade program quality and staff compensation, and higher qualification standards for Head Start teachers.

#### Family Workers in Head Start

More than 25,000 Family Workers are employed in local Head Start programs. Family Workers play a critical role in developing and supporting the implementation of Head Start's family partnership process. Through this process the family of each enrolled child has opportunities to develop and implement an individualized plan of services based on their interests and needs. In many instances, the quality of support received by families correlates with the training and qualifications of the program's Family Workers.

Collectively, Family Workers represent a group with varied levels of professional education/training and experience. Local agencies have established a range of qualifications for Family Workers varying from a Masters of Social Work (MSW) or other related degree to a High School diploma. Accordingly, some Head Start Social Services and Parent Involvement staff (known as Family and Community Partnerships staff since 1998) are college degreed as well as state licensed or credentialed. Others have received undergraduate training or on-the-job training. Some Family Workers are current or former parents of Head Start children. As might be expected, Family Workers also vary widely in characteristics such as salary levels, staffing patterns, fields of study, tenure, average salary, forms of supervision and

ongoing professional development opportunities.

Family Worker Training and Credentialing Initiative

The Family Worker Training and Credentialing Initiative is designed to implement a mandate from Congress in the Head Start Act Amendments of 1994 (P.L. 103–252). This Section of the Act required that "the Secretary, in coordination with concerned public and private agencies and organizations examining the issues of standards and training for family service workers, shall \* \* \* (1) review and, as necessary, revise or develop new qualification standards for Head Start staff providing such services; (2) promote the development of model curricula (on subjects including parenting training and family literacy) designed to ensure the attainment of appropriate competencies by individuals working or planning to work in the field of early childhood and family services; and (3) promote the establishment of a credential that indicates the attainment of the competencies that is accepted nationwide".

To assist in planning to carry out the Congressional mandates, the Head Start Bureau in 1999 and 2000 convened five focus groups of leaders from: national organizations, local Head Start Programs including parents of past and currently enrolled children; Head Start Quality Improvement Centers; accreditation organizations and higher education institutions to discuss the needs, issues, and existing models of Head Start Worker staff training.

Among the issues and needs identified by focus group participants were the following:

- Input to the development of "Competency Goals and Indicators for Head Start Staff Working with Families" attached in Appendix B, page 27. These competencies are being communicated to Head Start agencies to assist local efforts in selecting, training, and supervising Family Workers and will provide a common framework for competency-based training models solicited via this announcement;
- Key characteristics of models for delivery of training in these competencies and approaches to link competency-based training to higher education coursework and degrees;
- The importance and challenges of creating competency-based training that is responsive to highly diverse adult learners, such as Family Workers in Head Start and child care programs with little recent experience as students, who frequently continue to work while pursuing a degree, and who may require

special academic and social supports to successfully meet standards in general education and early childhood courses, and

• Recognition that with over twentynine percent of Head Start staff members being parents of former or current Head Start children that there is a necessity to ensure that any competency-based training program/ curriculum for Family Workers is appropriate for, open to, and welcomes the parents of Head Start children so that they can attain the necessary competencies.

Expectations and Requirements for Family Worker Training and Credentialing Projects

Section 649 the Head Start Act authorizes grants for research, demonstration and collaboration activities. These grants will involve extensive investigation into areas where knowledge is currently insufficient and will be awarded pursuant to Section 649.

Based on the above legislative mandates, focus group input, and additional planning, the Head Start Bureau is issuing this grants announcement to support the development and/or adaptation of a variety of models of competency-based, credit-bearing training for Family Workers in Head Start and early childhood and family support programs. The central requirements for all projects are as follows:

- Develop competency-based training programs and curricula relevant to the work of a Head Start Family Worker based on the Head Start Program Performance Standards; the Competency Goals and Indicators for Head Start Staff Working with Families, attached in Appendix B; and include a credible approach to assessing the attainment of these competencies by individual trainees;
- Create or adapt competency-based training that is linked to academic credit and degree programs and to other forms of credentialing for Family Workers. Applicants are urged to present plans for training which provide for articulation to AA, BA, MS degree programs if the trainee decides to continue his/her education, and portability, should trainees desire to be Family Workers in other related programs;
- Develop training and curricula that is accessible and affordable for adult learners and that accommodates the training needs of current Head Start Family Workers, including former Head Start Parents who are likely to continue to work full time as they continue to

participate in training towards a Family Worker credential; and

• Create state-of-the-art training and assessment strategies that will ultimately enhance the quality of program services and outcomes for the increasing diversity of low-income families served by Head Start and early childhood programs and agencies.

The Bureau is soliciting applications to develop and/or adapt competencybased training curricula and programs appropriate to the fulfillment of educational and professional growth needs of Family Worker staff nationwide, including all geographic regions as well as for staff serving Migrant and Indian families and communities. For purposes of this announcement "development" means the creation and design of a totally new competency-based, credit-based, training program. "Adaptation" means the proposed utilization of "as is" or slightly modified appropriate creditbearing competency-based training program coursework and materials, delivery modalities, scheduling and cost factors, etc.

Innovative, realistic, forward-looking, and trainee accessible model training program designs are necessary in order to facilitate and advance the Head Start Family Worker Training Initiative. Applicants may propose developmental work such as re-shaping course materials, curriculum and teaching strategies; adapting mentoring, advisement, reflective practice, and practicum strategies, using distance learning and other forms of technology in new ways, alternate means to improve access, reduce costs, and increase the successful completion of the training sequence and demonstration of competencies by candidates; and new efforts and methods to link competency-based training and curriculum to academic credit, higher education degree programs and related credentialing systems for Family Workers. Applicants are strongly encouraged to involve Family Workers, managers, program directors from Head Start and other community-based programs and training and technical assistance providers in their grant application planning and implementation of their projects. Attachment C provides a Directory of Head Start Training and Technical Assistance providers.

Each funded project will be expected to present a comprehensive competency-based credit-bearing training program and curriculum (plus alternate designs, if any) to the Head Start Bureau at the end of this grant project period. All elements of the

training program, including but not limited to recruitment, entry requirements, course content, credit hours, primary and alternate delivery modalities, time requirements, implementation plans and schedule, staffing qualifications, program and student assessments (including a method or strategy for the assessment of the competencies to be acquired by trainees), program accreditation, credentialing mechanisms, articulation plans/processes/agreements, and cost factors are to be included in this presentation. At some point in the future, the Head Start Bureau intends to require a common set of competencies and skills for Family Workers. Model curricula developed under this Announcement will be used to help determine the requisite training and credential attainment for these workers. Therefore, successful applicants are also expected to declare their intent to implement their proposed program after the end of the grant period, independent of any additional Federal support, if the Head Start Bureau determines that their model is sufficient to meet the training needs of Head Start Family Workers.

The Bureau in concert with national experts and practitioners will carry out a comprehensive review of all final submissions. The review will include examination of how proposed programs will enhance the capacities of trainees in all of the areas addressed in the Head Start Program Performance Standards and sub-areas of the eleven Family Workers "Competency Goals and Indicators" as defined in Appendix B. As a result of this review, the Bureau will examine the possible establishment and implementation of a Family Worker Training Program Resource Data Base incorporating all training programs conforming to Head Start's requirements. Those providers and programs included in the Data Base will be deemed to be responsive and appropriate for use by local program Family Worker staff in pursuing courses of studies and credentialing.

Grantees will be expected to attend a three-day Orientation Meeting regarding this Initiative in Washington D.C. to be held no later than six weeks after grant award. The Head Start Bureau and a work group of national consultants on competency-based training and credentialing will convene to engage with grantees regarding programmatic issues and Bureau expectations for this initiative. Applicants need to budget for the three-day Orientation meeting.

## Part II. Program Information and Requirements

#### A. Statutory Authority

The Head Start Act, as amended 42 U.S.C. 9801 et seq.

#### B. Eligible Applicants

Applicants must be public or private institutions of higher education or nonprofit or for profit organizations with experience and knowledge in working with early childhood programs for young children birth to age five. In accordance with 45 CFR 74.81, for profit organizations must waive their profit when applying for funding under this announcement.

#### C. Project Duration

Awards will be made on a competitive basis and will be for a oneyear period. The total project period will be one year.

#### D. Federal Share of Project Costs

A total of approximately \$1,000,000 in ACF funds will be available.

#### E. Number of Projects To Be Funded

ACF will fund up to ten applicants. An individual discretionary grant will be awarded to a successful applicant in order to foster achievement of the goals of this Head Start initiative.

#### F. Matching Requirement

Although there are no matching requirements, applicants are encouraged to provide non-Federal contributions to the project.

#### Part III. Application Requirements

#### A. Purpose

The project description provides a major means by which an application is evaluated and ranked to compete with other applications for available assistance. The project description should be concise and complete and should address the activity for which Federal funds are being requested. Supporting documents should be included where they can present information clearly and succinctly.

In preparing your project description, all information requested through each specific evaluation criteria should be provided. Awarding offices use this and other information in making their funding recommendations. It is important, therefore, that this information be included in the application.

#### B. General Instructions

ACF is particularly interested in specific factual information and statements of measurable goals in quantitative terms. Project descriptions are evaluated on the basis of substance, not length. Extensive exhibits are not required. Cross-referencing should be used rather than repetition. Supporting information concerning activities that will not be directly funded by the grant or information that does not directly pertain to an integral part of the grant funded activity should be placed in an appendix.

Pages should be numbered and a table of contents should be included for easy reference.

#### Introduction

Applicants are required to submit a full project description and shall prepare the project description statement in accordance with the following instructions and the specified evaluation criteria. The instructions give a broad overview of what your project description should include while the evaluation criteria expands and clarifies more program-specific information that is needed.

#### Project Summary/Abstract

Provide a summary of the Project description (a page or less) with reference to the funding request.

#### C. Objectives and Need for Assistance

Clearly identify the physical, economic, social, institutional and other problems(s) requiring a solution. The need for assistance must be demonstrated and the principal and subordinate objectives of the project must be clearly stated; supporting documentation, such as letters of support and testimonials from concerned interests other than the applicant, may be included. Any relevant data based on planning studies should be included or referred to in the endnotes/footnotes. Incorporate demographic data and participant/ beneficiary information, as needed. In developing the project description, the applicant may volunteer or be requested to provide information on the total range of projects currently being conducted and supported or (to be initiated), some of which may be outside the scope of the program announcement.

#### D. Results or Benefits Expected

Identify the results and benefits to be derived.

#### E. Approach

Outline a plan of action that describes the scope and detail of how the proposed work will be accomplished. Account for all functions or activities identified in the application. Cite factors that might accelerate or decelerate the work and state your reason for taking the proposed approach rather than others. Describe any unusual features of the project such as design or technological innovations, reductions in cost or time, or extraordinary social and community involvement.

Provide quantitative monthly or quarterly projections of the accomplishments to be achieved for each function or activity in such terms as the number of people to be served and the number of activities accomplished. When accomplishments cannot be quantified by activity or function, list them in chronological order to show the schedule of accomplishments and their target dates.

If any data is to be collected, maintained, and/or disseminated, clearance may be required from the U.S. Office of Management and Budget (OMB). This clearance pertains to any "collection of information that is conducted or sponsored by ACF."

List organizations, cooperating entities, consultants, or other key individuals whom will work on the project along with a short description of the nature of their effort or contribution.

#### F. Additional Information

Following are requests for additional information that need to be included in the application.

#### 1. Staff and Position Data

Provide a biographical sketch for each key person appointed and a job description for each vacant key position. A biographical sketch will also be required for new key staff as appointed.

#### 2. Organizational Profiles

Provide information on the applicant organization(s) and cooperating partners such as organizational charts, financial statements, audit reports or statements from CPAs/Licensed Public Accountants, Employer Identification Numbers, names of bond carriers, contact persons and telephone numbers, child care licenses and other documentation of professional accreditation, information on compliance with Federal/State/local government standards, documentation of experience in the program area, and other pertinent information.

#### G. Third-Party Agreements

Include written agreements between the grantee and subgrantees or subcontractors or other cooperative entities. These agreements must detail scope of work to be performed, work schedules, remuneration, and other terms and conditions that structure or define the relationship.

#### Letters of Support

Provide statements from community, public and commerical leaders that support the project proposed for funding. All submissions should be included in the application *OR* by application deadline.

#### H. Budget and Budget Justification

Provide line item detail and detailed calculations for each budget object class identified on the Budget Information form. Detailed calculations must include estimation methods, quantities, unit costs, and other similar quantitative detail sufficient for the calculation to be duplicated. The detailed budget must also include a breakout by the funding sources identified in Block 15 of the SF–424.

Provide a narrative budget justification that describes how the categorical costs are derived. Discuss the necessity, reasonableness, and allocability of the proposed costs.

#### General

The following guidelines are for preparing the budget and budget justification. Both Federal and non-Federal resources shall be detailed and justified in the budget and narrative justification. For purposes of preparing the budget and budget justification, "Federal resources" refers only to the ACF grant for which you are applying. Non-Federal resources are all other Federal and non-Federal resources. It is suggested that budget amounts and computations be presented in a columnar format: first column, object class categories; second column, Federal budget; next column(s), non-Federal budget(s), and last column, total budget. The budget justification should be a narrative.

#### Personnel

Description: Costs of employee salaries and wages.

Justification: Identify the project director or principal investigator, if known. For each staff person, provide the title, time commitment to the project (in months), time commitment to the project (as a percentage or full-time equivalent), annual salary, grant salary, wage rates, etc. Do not include the costs of consultants or personnel costs of delegate agencies or of specific project(s) or businesses to be financed by the applicant.

#### Fringe Benefits

*Description:* Costs of employee fringe benefits unless treated as part of an approved indirect cost rate.

Justification: Provide a breakdown of the amounts and percentages that comprise fringe benefit costs such as health insurance, FICA, retirement insurance, taxes, etc.

#### Travel

Description: Costs of project-related travel by employees of the applicant organization (does not include costs of consultant travel).

Justification: For each trip, show the total number of traveler(s), travel destination, duration of trip, per diem, mileage allowances, if privately owned vehicles will be used, and other transportation costs and subsistence allowances. Travel costs for key staff to attend ACF-sponsored workshops should be detailed in the budget.

#### Equipment

"Equipment" means an article of nonexpendable, tangible personal property having a useful life or more than one year and an acquisition cost which equals or exceeds the lesser of (a) the capitalization level established by the organization for the financial statement purposes, or (b) \$5,000. (Note: Acquisition cost means the net invoice unit price of an item of equipment, including the cost of any modifications, attachments, accessories, or auxiliary apparatus necessary to make it usable for the purpose for which it is acquired. Ancillary charges, such as taxes, duty, protective in-transit insurance, freight, and installation shall be included in or excluded from acquisition cost in accordance with the organization's regular written accounting practices.)

Justification: For each type of equipment requested, provide a description of the equipment, the cost per unit, the number of units, the total cost, and a plan for use on the project, as well as use or disposal of the equipment after the project ends. An applicant organization that uses its own definition for equipment should provide a copy of its policy or section of its policy which includes the equipment definition.

#### Supplies

Description: Costs of all tangible personal property other than that included under the Equipment category.

Justification: Specify general categories of supplies and their costs. Show computations and provide other information, which supports the amount requested.

#### Other

Enter the total of all other costs. Such costs, where applicable and appropriate, may include but are not limited to insurance, food, medical and dental costs (noncontractual), professional services costs, space and equipment rentals, printing and publication, computer use, training costs, such as tuition and stipends, staff development costs, and administrative costs.

*Justification:* Provide computations, a narrative description and a justification for each cost under this category.

#### **Indirect Charges**

Description: Total amount of indirect costs. This category should be used only when the applicant currently has an indirect cost rate approved by the Department of Health and Human Services (HHS) or another cognizant Federal agency.

*Justification:* An applicant that will charge indirect costs to the grant must enclose a copy of the current rate agreement. If the applicant organization is in the process of initially developing or renegotiating a rate, it should immediately upon notification that an award will be made, develop a tentative indirect cost rate proposal based on its most recently completed fiscal year in accordance with the principles set forth in the cognizant agency's guidelines for establishing indirect cost rates, and submit it to the cognizant agency. Applicants awaiting approval of their indirect cost proposals may also request indirect costs. It should be noted that when an indirect cost rate is requested, those costs included in the indirect cost pool should not also be charged as direct costs to the grant. Also, if the applicant is requesting a rate which is less than what is allowed under the program, the authorized representative of the applicant organization must submit a signed acknowledgement that the applicant is accepting a lower rate than allowed.

Total Direct Charges, Total Indirect Charges, Total Project Costs.

Self-explanatory.

#### **Part IV Evaluation Criteria**

#### A. Review Criteria

In considering how applicants will carry out the responsibilities addressed under this announcement, competing applications for financial assistance will be reviewed and evaluated against the following criteria:

Criterion 1. Objectives and Need for Assistance: (15 points)

The extent to which the application identifies relevant physical, economic, social, financial, institutional or other problems requiring a grant; demonstrates the need for assistance; states the principal and subordinate objectives of the project; provides supporting documentation or other testimonies from concerned interests other than the applicant.

Information provided in response to Part III, Section C. of this announcement will be used to evaluate applicants on this criterion.

Criterion 2. Results or Benefits Expected: (25 points)

The extent to which the application identifies the results and benefits to be derived; describes the anticipated contribution to policy, practice, theory and/or research; specific benefits should be described for Head Start and the whole early childhood community working with children birth through five.

The Head Start Bureau is particularly interested in the following:

1. Based on the stated program objectives, identify the results and benefits to be derived for Family Workers in Head Start, Early Head Start, and staff in other early childhood, child care and family support agencies.

2. Describe potential longer term benefits of this initiative, including enhancing participation and provision of higher education opportunities for family service staff; enhancement of relationships between higher education institutions and local early care and education programs, including Head Start; program quality, and practices and outcomes in early care child/family programs.

Information provided in response to Part III, Section D of this announcement will be used to evaluate applicants on this criterion.

#### Criterion 3. Approach: (50 points)

The extent to which the application outlines an acceptable plan of action pertaining to the scope of the project which details how the proposed work will be accomplished, including a timeline; lists of each organization, consultants, including the evaluator, or other key individuals who will work on the project along with a short description of the nature of their effort or contribution; assures the adequacy of time devoted to the project by key staff, the key staff should be knowledgeable of Head Start and Early Head Start, the applicant must fully describe the

approach and/or methodology and delineate the relationship of each task to the accomplishment of the proposed objectives. There should be evidence that the planned approach reflects sufficient input from collaborating

The Head Start Bureau is particularly interested in the following:

1. Describe the applicants' experience and capabilities in providing training to family worker staff from Head Start and early childhood and family support

programs.

Provide a discussion of the major current and emerging challenges facing family workers, and the challenges of delivering competency-based training to current staff members.

3. Describe the planning and development process the applicant will use to develop a final design/model program and describe how Head Start and other program family workers, managers and others will be involved.

- 4. Discuss how other career development and higher education organizations, institutions, and providers/partners or contributors may be involved in the planning and design phase, as well as in ongoing refinement and improvement of the desired model for curricula.
- 5. Propose and defend an initial overall professional development strategy for Head Start Family workers and other related early childhood higher education programs, including content, and sequence of development experience, and ways to encourage applications of new knowledge, standards and best practices to the instruction of participants and their sponsoring Head Start program. Include discussion of issues such as the admission/eligibility requirements, program scheduling, accessibility, and location of activities, including explicit approaches to supporting peer networking and mentoring of participants.

6. Provide assurance that training/ courses are offered at the lowest reasonable justifiable cost to trainees.

7. Indicate initial plans for the recruitment and selection of faculty or trainers who would train Family Workers. Discuss how recruitment and selection process will attract faculty/ trainers with demonstrated ability to respond to the growing diversity of the population of families and children served in Head Start, Early Head Start and other early care and education programs.

Information provided in response to Part III, Sections E, F and G of this announcement will be used to evaluate applicants on this criterion.

Criterion 4. Budget and Budget Justification: (10 points)

Provide line item detail and detailed calculations for each budget object class identified on the Budget Information form. Detailed calculations must include estimation methods, quantities, unit costs, and other similar quantitative detail sufficient for the calculation to be duplicated. The detailed budget must also include a breakout by the funding sources identified in Block 15 of the SF-424.

Provide a narrative budget justification that describes how the categorical costs are derived. Discuss the necessity, reasonableness and allocability of the proposed costs.

#### B. The Review Process

Applications received by the due date will be reviewed and scored competitively. Experts in the field, generally persons from outside the Federal government, will use the evaluation criteria listed in Part IV of this announcement to review and score the applications. The results of this review are a primary factor in making funding decisions. ACYF may also solicit comments from ACF Regional Office staff and other Federal agencies. The ACYF Commissioner may also consider a variety of all factors in funding decisions, including supporting a set of projects to serve Head Start programs and Family Workers in all geographic regions and representative of approaches to working with different types of Head Start programs, including Indian and Migrant grantees.

#### **Part V. The Application Process**

#### A. Required Forms

Eligible applicants interested in applying for funds must submit a complete application including the required forms included at the end of this program announcement in Appendix A. In order to be considered for a grant under this announcement, an application must be submitted on the Standard Form 424 approved by the Office of Management and Budget under Control Number 0348-0043. A copy has been provided. Each application must be signed by an individual authorized to act for the applicant and to assume responsibility for the obligations imposed by the terms and conditions of the grant award. Applicants requesting financial assistance for non-construction projects must file the Standard Form 424B, Assurances: Non-Construction Programs (approved by the Office of Management and Budget under control number 0348-0040). Applicants must sign and return the Standard Form 424B

with their application. Applicants must provide a certification concerning lobbying. Prior to receiving an award in excess of \$100,000, applicants shall furnish an executed copy of the lobbying certification (approved by the Office of Management and Budget under control number 0348–0046). Applicants must sign and return the certification with their application.

Applicants must make the appropriate certification of their compliance with the Drug-Free Workplace Act of 1988. By signing and submitting the application, applicants are providing the certification and need not mail back the certification with the application.

Applicants must make the appropriate certification that they are not presently debarred, suspended or otherwise ineligible for award. By signing and submitting the application, applicants are providing the certification and need not mail back the certification with the application.

Applicants must also understand that they will be held accountable for the smoking prohibition included within P.L. 103-227, Part C Environmental Tobacco Smoke (also known as The Pro-Children's Act of 1994). A copy of the Federal Register notice, which implements the smoking prohibition, is included with the forms. By signing and submitting the application, applicants are providing the certification and need not mail back the certification with the application.

#### B. Application Limits

The application should be doublespaced and single-sided on 81/2" × 11" plain white paper, with 1" margins on all sides. Use only a standard size font no smaller than 12 pitch throughout the application. All pages of the application (including appendices, resumes, charts, references/footnotes, tables, maps and exhibits) must be sequentially numbered, beginning on the first page after the budget justification, the principal investigator contact information and the Table of Contents. Although there is no limitation regarding number of pages, applicants are urged to be concise and limit applications to no more than 50 pages. Applicants are requested not to send pamphlets, brochures, or other printed material along with their applications as these pose copying difficulties. These materials, if submitted, will not be included in the review process. In addition, applicants must not submit any additional letters of endorsement beyond any that may be required.

Applicants are encouraged to submit curriculum vitae in a biographical format. Please note that applicants that do not comply with the requirements in the section on "Eligible Applicants" will not be included in the review process.

#### C. Checklist for a Complete Application

The checklist below is for your use to ensure that the application package has been properly prepared.

- One original, signed and dated application plus two copies.
- —Attachments/Appendices, when included, should be used only to provide supporting documentation such as resumes, and letters of agreement/support.
- (1) Application for Federal Assistance (SF–424, Rev. 7–97)
- (2) Budget information-non-construction programs (SF424A&B)
- (3) Budget Justification, including subcontract agency budgets
- (4) Application Narrative and Appendices
- (5) Proof that the organization is a non-profit organization
- (6) Assurances Non-Construction Program
  - (7) Certification Regarding Lobbying
- (8) If appropriate, a completed SPOC certification with the date of SPOC contact entered in line 16, page 1 of the SF-424, Rev. 7–97
- (9) Certification of Protection of Human Subjects

## D. Closing Date for Receipt of Applications

The closing time and date for receipt of applications is 5:00 p.m. (Eastern Time Zone) on August 1, 2001. Mailed or handcarried applications received after 5:00 p.m. on the closing date will be classified as late.

Deadline: Mailed applications shall be considered as meeting an announced deadline if they are received on or before the deadline time and date at the ACF Operations Center, 1815 North Fort Myer Drive, Suite 300, Arlington, Virginia 22209. Applicants are responsible for mailing applications well in advance when using all mail services to ensure that the applications are received on or before the deadline time and date.

Applications handcarried by applicants, applicant couriers, or other representatives of the applicant or by overnight/express mail couriers shall be considered as meeting an announced deadline if they are received on or before the deadline date, between the hours of 8:00 a.m. and 5:00 p.m. at the ACF Operations Center, 1815 North Fort Myer Drive, Suite 300, Arlington, VA 22209, between Monday and Friday (excluding Federal Holidays).

Applicants are cautioned that express/ overnight mail services may not always deliver as agreed.

ACF cannot accommodate the transmission of applications by FAX or through other electronic media. Therefore, applications transmitted to ACF electronically will not be accepted regardless of date or time of submission and time of receipt.

Late applications: Applications that do not meet the criteria stated above are considered late applications. ACF will notify each late applicant that its application will not be considered in the current competition.

Extension of deadlines: ACF may extend an application deadline for applicants affected by Acts of God such as floods and hurricanes, when there is widespread disruptions of mail service, or for other disruptions of services, such as a prolonged blackout, that affect the public at large. A determination to waive or to extend deadline requirements rests with the Chief Grants Management Officer.

#### E. Paperwork Reduction Act of 1995

Under the Paperwork Reduction Act of 1995, Public Law 104–13, the Department is required to submit to OMB for review and approval any reporting and record keeping requirements in regulations including program announcements. All information collections within this program announcement are approved under the following current valid OMB control numbers 0348–0043, 0348–0044, 03480–0040, 0348–0046, 0925–0418 and 0970–0139.

Public reporting burden for this collection is estimated to average 10 hours per response, including the time for reviewing instructions, gathering and maintaining the data needed and reviewing the collection of information.

An agency may not conduct or sponsor and a person is not required to respond to, a collection of information unless it displays a currently valid OMB control number.

#### F. Required Notification of the State Single Point of Contact

This program is covered under Executive Order 12372, Intergovernmental Review of Federal Programs, and 45 CFR part 100, Intergovernmental Review of Department of Health and Human Services Program and Activities. Under the Order, States may design their own processes for reviewing and commenting on proposed Federal assistance under covered programs.

\*All States and Territories except Alabama, Alaska, Arizona, Colorado,

Connecticut, Hawaii, Idaho, Indiana, Kansas, Louisiana, Massachusetts, Minnesota, Montana, Nebraska, New Jersey, New York, Ohio, Oklahoma, Oregon, Pennsylvania, South Dakota, Tennessee, Vermont, Virginia, Washington, Wyoming, and Palau have elected to participate in the Executive Order process and have established Single Points of Contact (SPOCs). Applicants from these twenty-seven jurisdictions need take no action regarding E.O. 12372. Applicants for projects to be administered by Federally-recognized Indian Tribes are also exempt from the requirements of E.O. 12372. Otherwise, applicants should contact their SPOCs as soon as possible to alert them of the prospective applications and receive any necessary instructions. Applicants must submit any required material to the SPOCs as soon as possible so that the program office can obtain and review SPOC comments as part of the award process. It is imperative that the applicant submit all required materials, if any, to the SPOC and indicate the date of this submittal (or the date of contact if no submittal is required) on the Standard Form 424, item 16a.

Under 45 CFR 100.8(a)(2), a SPOC has 60 days from the application deadline to comment on proposed new or competing continuation awards.

SPOCs are encouraged to eliminate the submission of routine endorsements as official recommendations.

Additionally, SPOCs are requested to clearly differentiate between mere advisory comments and those Official State process recommendations, which may trigger the accommodation or explain rule.

When comments are submitted directly to ACF, they should be addressed to William Wilson, Head Start Bureau, 330 C Street, SW, Washington, DC 20447, Attn: Head Start Family Worker Training and Credentialing Initiative. A list of Single Points of Contact for each State and Territory can be found on the web site: http://www.whitehouse.gov/omb/grants/spoc.html

(Catalog of Federal Domestic Program Number 93.600, Project Head Start)

Dated: April 26, 2001.

#### Gail E. Collins,

Acting Deputy Commissioner, Administration on Children, Youth and Families.

#### BILLING CODE 4184-01-P

Appendix A—Application Forms, Certifications, Disclosures, and Assurances

APPLICA <sup>*</sup>	TION FOR				OMB Approval No. 0348-004
FEDERAL	. ASSISTA	NCE	2. DATE SUBMITTED		Applicant Identifier
1. TYPE OF SU	BMISSION:		3. DATE RECEIVED B	YSTATE	State Application Identifier
Application Construc	tion	Preapplication Construction	4. DATE RECEIVED B	Y FEDERAL AGENCY	Federal Identifier
☐ Non-Con		Non-Construction			
5. APPLICANT	INFORMATION			10	4,7,7,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,
Legal Name:				Organizational Unit:	
Address (give c	ity, county, State	e, and zip code):		Name and telephone in this application (give a	number of person to be contacted on matters involving rea code)
6. EMPLOYER	IDENTIFICATIO	N NUMBER (EIN);		7. TYPE OF APPLICA	ANT: (enter appropriate letter in box)  H. Independent School Dist.
8. TYPE OF AP	PLICATION:			B. County	I. State Controlled Institution of Higher Learning
	☐ Nev	v Continuation	Revision	C. Municipal D. Township	J. Private University K. Indian Tribe
If Revision, ente	r appropriate let	ter(s) in box(es)		E. Interstate	L. Individual
				F. Intermunicipal	M. Profit Organization
A. Increase Award B. Decrease Award C. Increase Duration     D. Decrease Duration Other(specify):			e Duration	G. Special District	N. Other (Specify)
D. Declease L	ouration Other	specily).		9. NAME OF FEDERA	AL AGENCY:
<del></del>					
TITLE		OMESTIC ASSISTANCE N		11. DESCRIPTIVE TIT	TLE OF APPLICANT'S PROJECT:
13. PROPOSED	PROJECT	14. CONGRESSIONAL DI	STRICTS OF:		
Start Date	Ending Date	a. Applicant		b. Project	,
15. ESTIMATED	FUNDING:	1		16. IS APPLICATION ORDER 12372 PR	SUBJECT TO REVIEW BY STATE EXECUTIVE
a. Federal		\$	.00		APPLICATION/APPLICATION WAS MADE
b. Applicant	207.400	\$	.00	AVAILABLE	E TO THE STATE EXECUTIVE ORDER 12372 FOR REVIEW ON:
c. State		\$	.00	DATE	
d. Local		\$	.00	b. No. 🔲 PROGRA	M IS NOT COVERED BY E. O. 12372
e. Other \$		.00	OR PROGRAM HAS NOT BEEN SELECTED BY STAT		
f. Program Income \$ .00		00	17 IS THE ADDITION	NT DELINQUENT ON ANY FEDERAL DEBT?	
g. TOTAL		\$	.00	1	attach an explanation.
DOCUMENT H	AS BEEN DULY	AUTHORIZED BY THE GO	VERNING BODY OF TH		TON ARE TRUE AND CORRECT, THE HE APPLICANT WILL COMPLY WITH THE
		THE ASSISTANCE IS AWA			a Talanhara Nimahar
a. Type Name o	r Aumonzea Rep	oresemative	b. Title		c. Telephone Number
d. Signature of A	Authorized Repre	esentative			e. Date Signed

Previous Edition Usable Authorized for Local Reproduction Standard Form 424 (Rev. 7-97) Prescribed by OMB Circular A-102

#### **INSTRUCTIONS FOR THE SF-424**

Public reporting burden for this collection of information is estimated to average 45 minutes per response, including time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding the burden estimate or any other aspect of this collection of information, including suggestions for reducing this burden, to the Office of Management and Budget, Paperwork Reduction Project (0348-0043), Washington, DC 20503.

## PLEASE DO NOT RETURN YOUR COMPLETED FORM TO THE OFFICE OF MANAGEMENT AND BUDGET. SEND IT TO THE ADDRESS PROVIDED BY THE SPONSORING AGENCY.

This is a standard form used by applicants as a required facesheet for preapplications and applications submitted for Federal assistance. It will be used by Federal agencies to obtain applicant certification that States which have established a review and comment procedure in response to Executive Order 12372 and have selected the program to be included in their process, have been given an opportunity to review the applicant's submission.

 Item:
 Entry:

 1.
 Self-explanatory.

 Item:
 Entry:

 12.
 List only the largest political entities affected (e.g., State, counting, cities)

- Date application submitted to Federal agency (or State if applicable) and applicant's control number (if applicable).
- State use only (if applicable).
- If this application is to continue or revise an existing award, enter present Federal identifier number. If for a new project, leave blank.
- Legal name of applicant, name of primary organizational unit which will undertake the assistance activity, complete address of the applicant, and name and telephone number of the person to contact on matters related to this application.
- Enter Employer Identification Number (EIN) as assigned by the Internal Revenue Service.
- 7. Enter the appropriate letter in the space provided.
- 8. Check appropriate box and enter appropriate letter(s) in the space(s) provided:
  - -- "New" means a new assistance award.
  - "Continuation" means an extension for an additional funding/budget period for a project with a projected completion date.
  - "Revision" means any change in the Federal Government's financial obligation or contingent liability from an existing obligation.
- Name of Federal agency from which assistance is being requested with this application.
- Use the Catalog of Federal Domestic Assistance number and title of the program under which assistance is requested.
- 11. Enter a brief descriptive title of the project. If more than one program is involved, you should append an explanation on a separate sheet. If appropriate (e.g., construction or real property projects), attach a map showing project location. For preapplications, use a separate sheet to provide a summary description of this project.

- counties, cities).
- 13. Self-explanatory.
- List the applicant's Congressional District and any District(s) affected by the program or project.
- 15. Amount requested or to be contributed during the first funding/budget period by each contributor. Value of inkind contributions should be included on appropriate lines as applicable. If the action will result in a dollar change to an existing award, indicate only the amount of the change. For decreases, enclose the amounts in parentheses. If both basic and supplemental amounts are included, show breakdown on an attached sheet. For multiple program funding, use totals and show breakdown using same categories as item 15.
- Applicants should contact the State Single Point of Contact (SPOC) for Federal Executive Order 12372 to determine whether the application is subject to the State intergovernmental review process.
- 17. This question applies to the applicant organization, not the person who signs as the authorized representative. Categories of debt include delinquent audit disallowances, loans and taxes.
- 18. To be signed by the authorized representative of the applicant. A copy of the governing body's authorization for you to sign this application as official representative must be on file in the applicant's office. (Certain Federal agencies may require that this authorization be submitted as part of the application.)

SF-424 (Rev. 7-97) Back

		BUDGET INFORM SECT	IDGET INFORMATION - Non-Construction Programs SECTION A - BUDGET SUMMARY	struction Progran	ડા	OMB Approval No. 0348-0044
Grant Program Function	Catalog of Federal Domestic Assistance	Estimated Un	Estimated Unobligated Funds		New or Revised Budget	jet
or Activity	Number (b)	Federal	Non-Federal	Federal	Non-Federal	Total
1.	(2)	\$	\$	\$	\$	\$
23						
ю́.						
4.	The state of the s					
5. Totals		€	φ.	₩	₩	6
		SECTIC	SECTION B - BUDGET CATEGORIES	GORIES		
6 Object Class Categories	ries		GRANT PROGRAM, F	GRANT PROGRAM, FUNCTION OR ACTIVITY		Total
c. calcat class catego		(1)	(2)	(3)	(4)	(5)
a. Personnel		₩	<del>∨</del>	↔	₩.	€
b. Fringe Benefits	S					
c. Travel						
d. Equipment						
e. Supplies						
f. Contractual						
g. Construction						-
h. Other						
i. Total Direct Ch	i. Total Direct Charges (sum of 6a-6h)					
j. Indirect Charges	se					
k. TOTALS (sum of 6i and 6j)	n of 6i and 6j)	<del>⇔</del>	₩	€	€	€
7. Program Income		€ .	₩	€	€9	€9
Previous Edition Usable		Autho	Authorized for Local Reproduction	duction	Sr	Standard Form 424A (Rev. 7-97) Prescribed by OMB Circular A-102

Previous Edition Usable

	とうこうじゅ	VECTION C - NON-TRUBBAN ARMOGRACES	<b>用ひしし出し口り</b>		
(a) Grant Program		(b) Applicant	(c) State	(d) Other Sources	(e) TOTALS
8,		₩	\$	₩	₩.
Ġ					
10.					
12. TOTAL (sum of lines 8-11)		₩.	€9	49	4
	SECTION	SECTION D - FORECASTED CASH NEEDS	ASH NEEDS		
	Total for 1st Year	1st Quarter	2nd Quarter	3rd Quarter	4th Quarter
13. Federal	₩.	₩	€9	₩.	↔
14. Non-Federal					
15. TOTAL (sum of lines 13 and 14)	\$	₩.	€	49	↔
SECTION E - BUE	SECTION E - BUDGET ESTIMATES OF FEDERAL FUNDS NEEDED FOR BALANCE OF THE PROJECT	FEDERAL FUNDS NE	EDED FOR BALANCE	OF THE PROJECT	
(a) Grant Program			FUTURE FUNDIN	FUTURE FUNDING PERIODS (Years)	
		(b) First	(c) Second	(d) Third	(e) Fourth
16.		₩.	<u></u>	<del>. ⇔</del>	\$
17.					
18.					
19.					
20. TOTAL (sum of lines 16-19)		49	€9	₩.	4
	SECTION F	SECTION F - OTHER BUDGET INFORMATION	<b>FORMATION</b>		
21. Direct Charges:		22. Indired	22. Indirect Charges:		
23. Remarks:				And the state of t	
	Autho	Authorized for Local Reproduction	oduction	Standard F	Standard Form 424A (Rev. 7-97) Page

**Authorized for Local Reproduction** 

#### **INSTRUCTIONS FOR THE SF-424A**

Public reporting burden for this collection of information is estimated to average 180 minutes per response, including time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding the burden estimate or any other aspect of this collection of information, including suggestions for reducing this burden, to the Office of Management and Budget, Paperwork Reduction Project (0348-0044), Washington, DC 20503.

## PLEASE DO NOT RETURN YOUR COMPLETED FORM TO THE OFFICE OF MANAGEMENT AND BUDGET. SEND IT TO THE ADDRESS PROVIDED BY THE SPONSORING AGENCY.

#### **General Instructions**

This form is designed so that application can be made for funds from one or more grant programs. In preparing the budget, adhere to any existing Federal grantor agency guidelines which prescribe how and whether budgeted amounts should be separately shown for different functions or activities within the program. For some programs, grantor agencies may require budgets to be separately shown by function or activity. For other programs, grantor agencies may require a breakdown by function or activity. Sections A, B, C, and D should include budget estimates for the whole project except when applying for assistance which requires Federal authorization in annual or other funding period increments. In the latter case, Sections A, B, C, and D should provide the budget for the first budget period (usually a year) and Section E should present the need for Federal assistance in the subsequent budget periods. All applications should contain a breakdown by the object class categories shown in Lines a-k of Section B.

#### Section A. Budget Summary Lines 1-4 Columns (a) and (b)

For applications pertaining to a *single* Federal grant program (Federal Domestic Assistance Catalog number) and *not requiring* a functional or activity breakdown, enter on Line 1 under Column (a) the Catalog program title and the Catalog number in Column (b).

For applications pertaining to a *single* program *requiring* budget amounts by multiple functions or activities, enter the name of each activity or function on each line in Column (a), and enter the Catalog number in Column (b). For applications pertaining to multiple programs where none of the programs require a breakdown by function or activity, enter the Catalog program title on each line in *Column* (a) and the respective Catalog number on each line in Column (b).

For applications pertaining to *multiple* programs where one or more programs require a breakdown by function or activity, prepare a separate sheet for each program requiring the breakdown. Additional sheets should be used when one form does not provide adequate space for all breakdown of data required. However, when more than one sheet is used, the first page should provide the summary totals by programs.

#### Lines 1-4, Columns (c) through (g)

For new applications, leave Column (c) and (d) blank. For each line entry in Columns (a) and (b), enter in Columns (e), (f), and (g) the appropriate amounts of funds needed to support the project for the first funding period (usually a year).

For continuing grant program applications, submit these forms before the end of each funding period as required by the grantor agency. Enter in Columns (c) and (d) the estimated amounts of funds which will remain unobligated at the end of the grant funding period only if the Federal grantor agency instructions provide for this. Otherwise, leave these columns blank. Enter in columns (e) and (f) the amounts of funds needed for the upcoming period. The amount(s) in Column (g) should be the sum of amounts in Columns (e) and (f).

For supplemental grants and changes to existing grants, do not use Columns (c) and (d). Enter in Column (e) the amount of the increase or decrease of Federal funds and enter in Column (f) the amount of the increase or decrease of non-Federal funds. In Column (g) enter the new total budgeted amount (Federal and non-Federal) which includes the total previous authorized budgeted amounts plus or minus, as appropriate, the amounts shown in Columns (e) and (f). The amount(s) in Column (g) should not equal the sum of amounts in Columns (e) and (f).

Line 5 - Show the totals for all columns used.

#### **Section B Budget Categories**

In the column headings (1) through (4), enter the titles of the same programs, functions, and activities shown on Lines 1-4, Column (a), Section A. When additional sheets are prepared for Section A, provide similar column headings on each sheet. For each program, function or activity, fill in the total requirements for funds (both Federal and non-Federal) by object class categories.

Line 6a-i - Show the totals of Lines 6a to 6h in each column.

Line 6j - Show the amount of indirect cost.

Line 6k - Enter the total of amounts on Lines 6i and 6j. For all applications for new grants and continuation grants the total amount in column (5), Line 6k, should be the same as the total amount shown in Section A, Column (g), Line 5. For supplemental grants and changes to grants, the total amount of the increase or decrease as shown in Columns (1)-(4), Line 6k should be the same as the sum of the amounts in Section A, Columns (e) and (f) on Line 5.

Line 7 - Enter the estimated amount of income, if any, expected to be generated from this project. Do not add or subtract this amount from the total project amount, Show under the program

#### **INSTRUCTIONS FOR THE SF-424A** (continued)

narrative statement the nature and source of income. The estimated amount of program income may be considered by the Federal grantor agency in determining the total amount of the grant.

#### Section C. Non-Federal Resources

**Lines 8-11** Enter amounts of non-Federal resources that will be used on the grant. If in-kind contributions are included, provide a brief explanation on a separate sheet.

**Column (a)** - Enter the program titles identical to Column (a), Section A. A breakdown by function or activity is not necessary.

**Column (b)** - Enter the contribution to be made by the applicant.

**Column (c)** - Enter the amount of the State's cash and in-kind contribution if the applicant is not a State or State agency. Applicants which are a State or State agencies should leave this column blank.

**Column (d)** - Enter the amount of cash and in-kind contributions to be made from all other sources.

Column (e) - Enter totals of Columns (b), (c), and (d).

Line 12 - Enter the total for each of Columns (b)-(e). The amount in Column (e) should be equal to the amount on Line 5, Column (f), Section A.

#### Section D. Forecasted Cash Needs

Line 13 - Enter the amount of cash needed by quarter from the grantor agency during the first year.

Line 14 - Enter the amount of cash from all other sources needed by quarter during the first year.

Line 15 - Enter the totals of amounts on Lines 13 and 14.

## Section E. Budget Estimates of Federal Funds Needed for Balance of the Project

Lines 16-19 - Enter in Column (a) the same grant program titles shown in Column (a), Section A. A breakdown by function or activity is not necessary. For new applications and continuation grant applications, enter in the proper columns amounts of Federal funds which will be needed to complete the program or project over the succeeding funding periods (usually in years). This section need not be completed for revisions (amendments, changes, or supplements) to funds for the current year of existing grants.

If more than four lines are needed to list the program titles, submit additional schedules as necessary.

**Line 20** - Enter the total for each of the Columns (b)-(e). When additional schedules are prepared for this Section, annotate accordingly and show the overall totals on this line.

#### Section F. Other Budget Information

Line 21 - Use this space to explain amounts for individual direct object class cost categories that may appear to be out of the ordinary or to explain the details as required by the Federal grantor agency.

Line 22 - Enter the type of indirect rate (provisional, predetermined, final or fixed) that will be in effect during the funding period, the estimated amount of the base to which the rate is applied, and the total indirect expense.

Line 23 - Provide any other explanations or comments deemed necessary.

OMB Approval No. 0348-0040

#### **ASSURANCES - NON-CONSTRUCTION PROGRAMS**

Public reporting burden for this collection of information is estimated to average 15 minutes per response, including time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding the burden estimate or any other aspect of this collection of information, including suggestions for reducing this burden, to the Office of Management and Budget, Paperwork Reduction Project (0348-0040), Washington, DC 20503.

## PLEASE DO NOT RETURN YOUR COMPLETED FORM TO THE OFFICE OF MANAGEMENT AND BUDGET. SEND IT TO THE ADDRESS PROVIDED BY THE SPONSORING AGENCY.

NOTE: Certain of these assurances may not be applicable to your project or program. If you have questions, please contact the awarding agency. Further, certain Federal awarding agencies may require applicants to certify to additional assurances. If such is the case, you will be notified.

As the duly authorized representative of the applicant, I certify that the applicant:

- Has the legal authority to apply for Federal assistance and the institutional, managerial and financial capability (including funds sufficient to pay the non-Federal share of project cost) to ensure proper planning, management and completion of the project described in this application.
- Will give the awarding agency, the Comptroller General of the United States and, if appropriate, the State, through any authorized representative, access to and the right to examine all records, books, papers, or documents related to the award; and will establish a proper accounting system in accordance with generally accepted accounting standards or agency directives.
- Will establish safeguards to prohibit employees from using their positions for a purpose that constitutes-or presents the appearance of personal or organizational conflict of interest, or personal gain.
- Will initiate and complete the work within the applicable time frame after receipt of approval of the awarding agency.
- Will comply with the Intergovernmental Personnel Act of 1970 (42 U.S.C. §§4728-4763) relating to prescribed standards for merit systems for programs funded under one of the 19 statutes or regulations specified in Appendix A of OPM's Standards for a Merit System of Personnel Administration (5 C.F.R. 900, Subpart F).
- 6. Will comply with all Federal statutes relating to nondiscrimination. These include but are not limited to: (a) Title VI of the Civil Rights Act of 1964 (P.L. 88-352) which prohibits discrimination on the basis of race, color or national origin; (b) Title IX of the Education Amendments of 1972, as amended (20 U.S.C. §§1681-1683, and 1685-1686), which prohibits discrimination on the basis of sex; (c) Section 504 of the Rehabilitation

- Act of 1973, as amended (29 U.S.C. §794), which prohibits discrimination on the basis of handicaps; (d) the Age Discrimination Act of 1975, as amended (42 U.S.C. §§6101-6107), which prohibits discrimination on the basis of age; (e) the Drug Abuse Office and Treatment Act of 1972 (P.L. 92-255), as amended, relating to nondiscrimination on the basis of drug abuse: (f) the Comprehensive Alcohol Abuse and Alcoholism Prevention, Treatment and Rehabilitation Act of 1970 (P.L. 91-616), as amended, relating to nondiscrimination on the basis of alcohol abuse or alcoholism; (g) §§523 and 527 of the Public Health Service Act of 1912 (42 U.S.C. §§290 dd-3 and 290 ee 3), as amended, relating to confidentiality of alcohol and drug abuse patient records; (h) Title VIII of the Civil Rights Act of 1968 (42 U.S.C. §§3601 et seq.), as amended, relating to nondiscrimination in the sale, rental or financing of housing; (i) any other nondiscrimination provisions in the specific statute(s) under which application for Federal assistance is being made; and, (j) the requirements of any other nondiscrimination statute(s) which may apply to the application.
- 7. Will comply, or has already complied, with the requirements of Titles II and III of the Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970 (P.L. 91-646) which provide for fair and equitable treatment of persons displaced or whose property is acquired as a result of Federal or federally-assisted programs. These requirements apply to all interests in real property acquired for project purposes regardless of Federal participation in purchases.
- Will comply, as applicable, with provisions of the Hatch Act (5 U.S.C. §§1501-1508 and 7324-7328) which limit the political activities of employees whose principal employment activities are funded in whole or in part with Federal funds.

- 9. Will comply, as applicable, with the provisions of the Davis-Bacon Act (40 U.S.C. §§276a to 276a-7), the Copeland Act (40 U.S.C. §276c and 18 U.S.C. §874), and the Contract Work Hours and Safety Standards Act (40 U.S.C. §§327-333), regarding labor standards for federally-assisted construction subagreements.
- 10. Will comply, if applicable, with flood insurance purchase requirements of Section 102(a) of the Flood Disaster Protection Act of 1973 (P.L. 93-234) which requires recipients in a special flood hazard area to participate in the program and to purchase flood insurance if the total cost of insurable construction and acquisition is \$10,000 or more.
- 11. Will comply with environmental standards which may be prescribed pursuant to the following: (a) institution of environmental quality control measures under the National Environmental Policy Act of 1969 (P.L. 91-190) and Executive Order (EO) 11514; (b) notification of violating facilities pursuant to EO 11738; (c) protection of wetlands pursuant to EO 11990; (d) evaluation of flood hazards in floodplains in accordance with EO 11988; (e) assurance of project consistency with the approved State management program developed under the Coastal Zone Management Act of 1972 (16 U.S.C. §§1451 et seq.); (f) conformity of Federal actions to State (Clean Air) Implementation Plans under Section 176(c) of the Clean Air Act of 1955, as amended (42 U.S.C. §§7401 et seq.); (g) protection of underground sources of drinking water under the Safe Drinking Water Act of 1974, as amended (P.L. 93-523); and, (h) protection of endangered species under the Endangered Species Act of 1973, as amended (P.L. 93-205).

- Will comply with the Wild and Scenic Rivers Act of 1968 (16 U.S.C. §§1271 et seq.) related to protecting components or potential components of the national wild and scenic rivers system.
- 13. Will assist the awarding agency in assuring compliance with Section 106 of the National Historic Preservation Act of 1966, as amended (16 U.S.C. §470), EO 11593 (identification and protection of historic properties), and the Archaeological and Historic Preservation Act of 1974 (16 U.S.C. §§469a-1 et seq.).
- Will comply with P.L. 93-348 regarding the protection of human subjects involved in research, development, and related activities supported by this award of assistance.
- 15. Will comply with the Laboratory Animal Welfare Act of 1966 (P.L. 89-544, as amended, 7 U.S.C. §§2131 et seq.) pertaining to the care, handling, and treatment of warm blooded animals held for research, teaching, or other activities supported by this award of assistance.
- Will comply with the Lead-Based Paint Poisoning Prevention Act (42 U.S.C. §§4801 et seq.) which prohibits the use of lead-based paint in construction or rehabilitation of residence structures.
- 17. Will cause to be performed the required financial and compliance audits in accordance with the Single Audit Act Amendments of 1996 and OMB Circular No. A-133, "Audits of States, Local Governments, and Non-Profit Organizations."
- Will comply with all applicable requirements of all other Federal laws, executive orders, regulations, and policies governing this program.

SIGNATURE OF AUTHORIZED CERTIFYING OFFICIAL	TITLE		
APPLICANT ORGANIZATION		DATE SUBMITTED	

## Administration for Children, Youth, and Families U.S. Department of Health and Human Services

#### CERTIFICATION REGARDING DRUG-FREE WORKPLACE REQUIREMENTS

This certification is required by the regulations implementing the Drug-Free Workplace Act of 1988: 45 CFR Part 76. Subpart, F. Sections 76.630(c) and (d)(2) and 76.645(a)(1) and (b) provide that a Federal agency may designate a central receipt point for STATE-WIDE AND STATE AGENCY-WIDE certifications, and for notification of criminal drug convictions. For the Department of Health and Human Services, the central pint is: Division of Grants Management and Oversight. Office of Management and Acquisition, Department of Health and Human Services. Room 517-D. 200 Independence Avenue, SW Washington, DC 20201.

Certification Regarding Drug-Free Workplace Requirements (Instructions for Certification)

- 1. By signing and/or submitting this application or grant agreement, the grantee is providing the certification set out below.
- 2. The certification set out below is a material representation of fact upon which reliance is placed when the agency awards the grant. If it is later determined that the grantee knowingly rendered a false certification, or otherwise violates the requirements of the Drug-Free Workplace Act, the agency, in addition to any other remedies available to the Federal Government, may take action authorized under the Drug-Free Workplace Act.
- 3. For grantees other than individuals. Alternate I applies.
- 4. For grantees who are individuals, Alternate II applies.
- 5. Workplaces under grants, for grantees other than individuals, need not be identified on the certification. If known, they may be identified in the grant application. If the grantee does not identify the workplaces at the time of application, or upon award, if there is no application, the grantee must keep the identity of the workplace(s) on file in its office and make the information available for Federal inspection. Failure to identify all known workplaces constitutes a violation of the grantee's drug-free workplace requirements.
- 6. Workplace identifications must include the actual address of buildings (or parts of buildings) or other sites where work under the grant takes place. Categorical descriptions may be used (e.g., all vehicles of a mass transit authority or State highway department while in operation, State employees in each local unemployment office, performers in concert halls or radio studios).
- 7. If the workplace identified to the agency changes during the performance of the grant, the grantee shall inform the agency of the change(s), if it previously identified the workplaces in question (see paragraph five).
- 8. Definitions of terms in the Nonprocurement Suspension and Debarment common rule and Drug-Free Workplace common rule apply to this certification. Grantees' attention is called, in particular, to the following definitions from these rules:

Controlled substance means a controlled substance in Schedules I through V of the Controlled Substances Act (21 U.S.C. 812) and as further defined by regulation (21 CFR 1308.11 through 1308.15);

Conviction means a finding of guilt (including a plea of nolo contendere) or imposition of sentence, or both, by any judicial body charged with the responsibility to determine violations of the Federal or State criminal drug statutes;

Criminal drug statute means a Federal or non-Federal criminal statute involving the manufacture, distribution, dispensing, use, or possession of any controlled substance;

Employee means the employee of a grantee directly engaged in the performance of work under a grant, including: (i) All direct charge employees; (ii) All indirect charge employees unless their impact or involvement is insignificant to

the performance of the grant; and, (iii) Temporary personnel and consultants who are directly engaged in the performance of work under the grant and who are on the grantee's payroll. This definition does not include workers not on the payroll of the grantee (e.g., volunteers, even if used to meet a matching requirement; consultants or independent contractors not on the grantee's payroll; or employees of subrecipients or subcontractors in covered workplaces).

#### Certification Regarding Drug-Free Workplace Requirements

Alternate I. (Grantees Other Than Individuals)

The grantee certifies that it will or will continue to provide a drug-free workplace by:

- (a) Publishing a statement notifying employees that the unlawful manufacture, distribution, dispensing, possession, or use of a controlled substance is prohibited in the grantee's workplace and specifying the actions that will be taken against employees for violation of such prohibition.
- (b) Establishing an ongoing drug-free awareness program to inform employees about -- (1)The dangers of drug abuse in the workplace: (2) The grantee's policy of maintaining a drug-free workplace; (3) Any available drug counseling, rehabilitation, and employee assistance programs; and (4) The penalties that may be imposed upon employees for drug abuse violations occurring in the workplace:
- (c) Making it a requirement that each employee to be engaged in the performance of the grant be given a copy of the statement required by paragraph (a);
- (d) Notifying the employee in the statement required by paragraph (a) that, as a condition of employment under the grant, the employee will -- (1) Abide by the terms of the statement; and (2) Notify the employer in writing of his or her conviction for a violation of a criminal drug statute occurring in the workplace no later than five calendar days after such conviction:
- (e) Notifying the agency in writing, within ten calendar days after receiving notice under paragraph (d)(2) from an employee or otherwise receiving actual notice of such conviction. Employers of convicted employees must provide notice, including position title, to every grant officer or other designee on whose grant activity the convicted employee was working, unless the Federal agency has designated a central point for the receipt of such notices. Notice shall include the identification number(s) of each affected grant:
- (f) Taking one of the following actions, within 30 calendar days of receiving notice under paragraph (d)(2), with respect to any employee who is so convicted (1) Taking appropriate personnel action against such an employee, up to and including termination, consistent with the requirements of the Rehabilitation Act of 1973, as amended; or (2) Requiring such employee to participate satisfactorily in a drug abuse assistance or rehabilitation program approved for such purposes by a Federal, State, or local health, law enforcement, or other appropriate agency;
- (g) Making a good faith effort to continue to maintain a drug-free workplace through implementation of paragraphs (a), (b), (c), (d), (e) and (f).
- (B) The grantee may insert in the space provided below the site(s) for the performance of work done in connection with the specific grant:

Place of Performance (Street address, city, county, state, zip code)
Check if there are workplaces on file that are not identified here.

#### Alternate II. (Grantees Who Are Individuals)

- (a) The grantee certifies that, as a condition of the grant, he or she will not engage in the unlawful manufacture, distribution, dispensing, possession, or use of a controlled substance in conducting any activity with the grant;
- (b) If convicted of a criminal drug offense resulting from a violation occurring during the conduct of any grant activity, he or she will report the conviction, in writing, within 10 calendar days of the conviction, to every grant officer or other designee, unless the Federal agency designates a central point for the receipt of such notices. When notice is made to such a central point, it shall include the identification number(s) of each affected grant.

[55 FR 21690, 21702, May 25, 1990]

## Administration for Children, Youth, and Families U.S. Department of Health and Human Services

#### CERTIFICATION REGARDING DEBARMENT, SUSPENSION AND OTHER RESPONSIBILITY MATTERS

Certification Regarding Debarment, Suspension, and Other Responsibility Matters--Primary Covered Transactions

#### Instructions for Certification

- 1. By signing and submitting this proposal, the prospective primary participant is providing the certification set out below.
- 2. The inability of a person to provide the certification required below will not necessarily result in denial of participation in this covered transaction. The prospective participant shall submit an explanation of why it cannot provide the certification set out below. The certification or explanation will be considered in connection with the department or agency's determination whether to enter into this transaction. However, failure of the prospective primary participant to furnish a certification or an explanation shall disqualify such person from participation in this transaction.
- 3. The certification in this clause is a material representation of fact upon which reliance was placed when the department or agency determined to enter into this transaction. If it is later determined that the prospective primary participant knowingly rendered an erroneous certification, in addition to other remedies available to the Federal Government, the department or agency may terminate this transaction for cause or default.
- 4. The prospective primary participant shall provide immediate written notice to the department or agency to which this proposal is submitted if at any time the prospective primary participant learns that its certification was erroneous when submitted or has become erroneous by reason of changed circumstances.
- 5. The terms covered transaction, debarred, suspended, ineligible, lower tier covered transaction, participant, person, primary covered transaction, principal, proposal, and voluntarily excluded, as used in this clause, have the meanings set out in the Definitions and Coverage sections of the rules implementing Executive Order 12549. You may contact the department or agency to which this proposal is being submitted for assistance in obtaining a copy of those regulations.
- 6. The prospective primary participant agrees by submitting this proposal that, should the proposed covered transaction be entered into, it shall not knowingly enter into any lower tier covered transaction with a person who is proposed for debarment under 48 CFR part 9, subpart 9.4, debarred, suspended, declared ineligible, or voluntarily excluded from participation in this covered transaction, unless authorized by the department or agency entering into this transaction.
- 7. The prospective primary participant further agrees by submitting this proposal that it will include the clause titled "Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion-Lower Tier Covered Transaction," provided by the department or agency entering into this covered transaction, without modification, in all lower tier covered transactions and in all solicitations for lower tier covered transactions.
- 8. A participant in a covered transaction may rely upon a certification of a prospective participant in a lower tier covered transaction that it is not proposed for debarment under 48 CFR part 9, subpart 9.4, debarred, suspended, ineligible, or voluntarily excluded from the covered transaction, unless it knows that the certification is erroneous. A participant may decide the method and frequency by which it determines the eligibility of its principals. Each participant may, but is not required to, check the List of Parties Excluded from Federal Procurement and

Nonprocurement Programs.

9. Nothing contained in the foregoing shall be construed to require establishment of a system of records in order to render in good faith the certification required by this clause. The knowledge and information of a participant is not required to exceed that which is normally possessed by a prudent person in the ordinary course of business dealings. 10. Except for transactions authorized under paragraph 6 of these instructions, if a participant in a covered transaction knowingly enters into a lower tier covered transaction with a person who is proposed for debarment under 48 CFR part 9, subpart 9.4, suspended, debarred, ineligible, or voluntarily excluded from participation in this transaction, in addition to other remedies available to the Federal Government, the department or agency may terminate this transaction for cause or default.

\* \* \* \* \*

Certification Regarding Debarment, Suspension, and Other Responsibility Matters--Primary Covered Transactions

- (1) The prospective primary participant certifies to the best of its knowledge and belief, that it and its principals:
- (a) Are not presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded by any Federal department or agency:
- (b) Have not within a three-year period preceding this proposal been convicted of or had a civil judgment rendered against them for commission of fraud or a criminal offense in connection with obtaining, attempting to obtain, or performing a public (Federal, State or local) transaction or contract under a public transaction; violation of Federal or State antitrust statutes or commission of embezzlement, theft, forgery, bribery, falsification or destruction of records, making false statements, or receiving stolen property;
- (c) Are not presently indicted for or otherwise criminally or civilly charged by a governmental entity (Federal, State or local) with commission of any of the offenses enumerated in paragraph (1)(b) of this certification; and
- (d) Have not within a three-year period preceding this application/proposal had one or more public transactions (Federal. State or local) terminated for cause or default.
- (2) Where the prospective primary participant is unable to certify to any of the statements in this certification, such prospective participant shall attach an explanation to this proposal.

Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion--Lower Tier Covered Transactions

#### Instructions for Certification

- 1. By signing and submitting this proposal, the prospective lower tier participant is providing the certification set out below.
- 2. The certification in this clause is a material representation of fact upon which reliance was placed when this transaction was entered into. If it is later determined that the prospective lower tier participant knowingly rendered an erroneous certification, in addition to other remedies available to the Federal Government the department or agency with which this transaction originated may pursue available remedies, including suspension and/or debarment.
- 3. The prospective lower tier participant shall provide immediate written notice to the person to which this proposal is submitted if at any time the prospective lower tier participant learns that its certification was erroneous when submitted or had become erroneous by reason of changed circumstances.
- 4. The terms covered transaction, debarred, suspended, ineligible, lower tier covered transaction, participant, person, primary covered transaction, principal, proposal, and voluntarily excluded, as used in this clause, have the meaning set out in the Definitions and Coverage sections of rules implementing Executive Order 12549. You may contact the person to which this proposal is submitted for assistance in obtaining a copy of those regulations.
- 5. The prospective lower tier participant agrees by submitting this proposal that, should the proposed covered transaction be entered into, it shall not knowingly enter into any lower tier covered transaction with a person who is proposed for debarment under 48 CFR part 9, subpart 9.4, debarred, suspended, declared ineligible, or voluntarily excluded from participation in this covered transaction, unless authorized by the department or agency with which this transaction originated.
- 6. The prospective lower tier participant further agrees by submitting this proposal that it will include this clause titled "Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion-Lower Tier Covered

Transaction," without modification, in all lower tier covered transactions and in all solicitations for lower tier covered transactions.

- 7. A participant in a covered transaction may rely upon a certification of a prospective participant in a lower tier covered transaction that it is not proposed for debarment under 48 CFR part 9, subpart 9.4, debarred, suspended, ineligible, or voluntarily excluded from covered transactions, unless it knows that the certification is erroneous. A participant may decide the method and frequency by which it determines the eligibility of its principals. Each participant may, but is not required to, check the List of Parties Excluded from Federal Procurement and Nonprocurement Programs.
- 8. Nothing contained in the foregoing shall be construed to require establishment of a system of records in order to render in good faith the certification required by this clause. The knowledge and information of a participant is not required to exceed that which is normally possessed by a prudent person in the ordinary course of business dealings.

  9. Except for transactions authorized under paragraph 5 of these instructions, if a participant in a covered transaction knowingly enters into a lower tier covered transaction with a person who is proposed for debarment under 48 CFR part 9, subpart 9.4, suspended, debarred, ineligible, or voluntarily excluded from participation in this transaction, in addition to other remedies available to the Federal Government, the department or agency with which this transaction originated may pursue available remedies, including suspension and/or debarment.

\* \* \* \* \*

Certification Regarding Debarment, Suspension. Ineligibility and Voluntary Exclusion--Lower Tier Covered Transactions

- (1) The prospective lower tier participant certifies, by submission of this proposal, that neither it nor its principals is presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from participation in this transaction by any Federal department or agency.
- (2) Where the prospective lower tier participant is unable to certify to any of the statements in this certification, such prospective participant shall attach an explanation to this proposal.

## Administration for Children, Youth, and Families U.S. Department of Health and Human Services

#### CERTIFICATION REGARDING LOBBYING

Certification for Contracts, Grants, Loans, and Cooperative Agreements

The undersigned certifies, to the best of his or her knowledge and belief, that:

- (1) No Federal appropriated funds have been paid or will be paid, by or on behalf of the undersigned, to any person for influencing or attempting to influence an officer or employee of an agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with the awarding of any Federal contract, the making of any Federal grant, the making of any Federal loan, the entering into of any cooperative agreement, and the extension, continuation, renewal, amendment, or modification of any Federal contract, grant, loan, or cooperative agreement.
- (2) If any funds other than Federal appropriated funds have been paid or will be paid to any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with this Federal contract, grant, loan, or cooperative agreement, the undersigned shall complete and submit Standard Form-LLL, "Disclosure Form to Report Lobbying," in accordance with its instructions.
- (3) The undersigned shall require that the language of this certification be included in the award documents for all subawards at all tiers (including subcontracts, subgrants, and contracts under grants, loans, and cooperative agreements) and that all subrecipients shall certify and disclose accordingly. This certification is a material representation of fact upon which reliance was placed when this transaction was made or entered into. Submission of this certification is a prerequisite for making or entering into this transaction imposed by section 1352, title 31, U.S. Code. Any person who fails to file the required certification shall be subject to a civil penalty of not less than \$10,000 and not more than \$100,000 for each such failure.

Statement for Loan Guarantees and Loan Insurance

The undersigned states, to the best of his or her knowledge and belief, that:

If any funds have been paid or will be paid to any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with this commitment providing for the United States to insure or guarantee a loan, the undersigned shall complete and submit Standard Form-LLL, "Disclosure Form to Report Lobbying," in accordance with its instructions. Submission of this statement is a prerequisite for making or entering into this transaction imposed by section 1352, title 31, U.S. Code. Any person who fails to file the required statement shall be subject to a civil penalty of not less than \$10,000 and not more than \$100,000 for each such failure.

Signature					***************************************	
Title	·····					•••••
Organization	••••••	•••••	·····	••••••	***********	<del></del>

### **DISCLOSURE OF LOBBYING ACTIVITIES**

Complete this form to disclose lobbying activities pursuant to 31 U.S.C. 1352

Approved by OMB 0348-0046

	(See reverse for pu	ıblic burden disclosu	re.)	
1. Type of Federal Action:  a. contract b. grant c. cooperative agreement d. loan e. loan guarantee f. loan insurance  4. Name and Address of Reporting Prime Subawardee Tier	al Action: offer/application al award -award	a. initial filing b. material Characterial Characteria	nange	
Congressional District, if known 6. Federal Department/Agency:	:		District, <i>if known</i> : m Name/Description:	
8. Federal Action Number, if known	CFDA Number, A	if known:		
o. rederal Action Number, // knowl	7.	9. Award Amount	, II KIIOWII:	
10. a. Name and Address of Lobby (if individual, last name, first n	•	b. Individuals Per different from N (last name, first	,	cluding address if
11. Information requested through this form is authorized 1352. This disclosure of lobbying activities is a ma upon which reliance was placed by the tier above when or entered into. This disclosure is required pursuan information will be reported to the Congress semi-annu public inspection. Any person who fails to file the r subject to a civil penalty of not less that \$10,000 and each such failure.	Print Name:			
Federal Use Only:			Au	othorized for Local Reproduction andard Form LLL (Rev. 7-97)

#### INSTRUCTIONS FOR COMPLETION OF SF-LLL, DISCLOSURE OF LOBBYING ACTIVITIES

This disclosure form shall be completed by the reporting entity, whether subawardee or prime Federal recipient, at the initiation or receipt of a covered Federal action, or a material change to a previous filing, pursuant to title 31 U.S.C. section 1352. The filing of a form is required for each payment or agreement to make payment to any lobbying entity for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with a covered Federal action. Complete all items that apply for both the initial filing and material change report. Refer to the implementing guidance published by the Office of Management and Budget for additional information.

- 1. Identify the type of covered Federal action for which lobbying activity is and/or has been secured to influence the outcome of a covered Federal action.
- 2. Identify the status of the covered Federal action.
- Identify the appropriate classification of this report. If this is a followup report caused by a material change to the information previously reported, enter
  the year and quarter in which the change occurred. Enter the date of the last previously submitted report by this reporting entity for this covered Federal
  action.
- 4. Enter the full name, address, city, State and zip code of the reporting entity. Include Congressional District, if known. Check the appropriate classification of the reporting entity that designates if it is, or expects to be, a prime or subaward recipient. Identify the tier of the subawardee, e.g., the first subawardee of the prime is the 1st tier. Subawards include but are not limited to subcontracts, subgrants and contract awards under grants.
- 5. If the organization filing the report in item 4 checks "Subawardee," then enter the full name, address, city, State and zip code of the prime Federal recipient. Include Congressional District, if known.
- 6. Enter the name of the Federal agency making the award or loan commitment. Include at least one organizationallevel below agency name, if known. For example, Department of Transportation, United States Coast Guard.
- 7. Enter the Federal program name or description for the covered Federal action (item 1). If known, enter the full Catalog of Federal Domestic Assistance (CFDA) number for grants, cooperative agreements, loans, and loan commitments.
- 8. Enter the most appropriate Federal identifying number available for the Federal action identified in item 1 (e.g., Request for Proposal (RFP) number; Invitation for Bid (IFB) number; grant announcement number; the contract, grant, or loan award number; the application/proposal control number assigned by the Federal agency). Include prefixes, e.g., "RFP-DE-90-001."
- 9. For a covered Federal action where there has been an award or loan commitment by the Federal agency, enter the Federal amount of the award/loan commitment for the prime entity identified in item 4 or 5.
- 10. (a) Enter the full name, address, city, State and zip code of the lobbying registrant under the Lobbying Disclosure Act of 1995 engaged by the reporting entity identified in item 4 to influence the covered Federal action.
  - (b) Enter the full names of the individual(s) performing services, and include full address if different from 10 (a). Enter Last Name, First Name, and Middle Initial (MI).
- 11. The certifying official shall sign and date the form, print his/her name, title, and telephone number.

According to the Paperwork Reduction Act, as amended, no persons are required to respond to a collection of information unless it displays a valid OMB Control Number. The valid OMB control number for this information collection is OMB No. 0348-0046. Public reporting burden for this collection of information is estimated to average 10 minutes per response, including time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding the burden estimate or any other aspect of this collection of information, including suggestions for reducing this burden, to the Office of Management and Budget, Paperwork Reduction Project (0348-0046), Washington, DC 20503.

#### CERTIFICATION REGARDING ENVIRONMENTAL TOBACCO SMOKE

Public Law 103227, Part C Environmental Tobacco Smoke, also known as the Pro Children Act of 1994, requires that smoking not be permitted in any portion of any indoor routinely owned or leased or contracted for by an entity and used routinely or regularly for provision of health, day care, education, or library services to children under the age of 18, if the services are funded by Federal programs either directly or through State or local governments, by Federal grant, contract, loan, or loan guarantee. The law does not apply to children's services provided in private residences, facilities funded solely by Medicare or Medicaid funds, and portions of facilities used for inpatient drug or alcohol treatment. Failure to comply with the provisions of the law may result in the imposition of a civil monetary penalty of up to \$1000 per day and/or the imposition of an administrative compliance order on the responsible entity. By signing and submitting this application the applicant/grantee certifies that it will comply with the requirements of the Act.

The applicant/grantee further agrees that it will require the language of this certification be included in any subawards which contain provisions for the children's services and that all subgrantees shall certify accordingly.

#### BILLING CODE 4184-01-C

It is estimated that in 2001 the Federal Government will outlay \$305.6 billion in grants to State and local governments. Executive Order 12372, "Intergovernmental Review of Federal Programs" was issued with the desire to foster the intergovernmental partnership and strengthen federalism by relying on State and local processes for the coordination and review of proposed Federal financial assistance and direct Federal development. The Order allows each State to designate an entity to perform this function. Below is the official list of those entities. For those States that have a home page for their designated entity, a direct link has been provided below. States that are not listed on this page have chosen not to participate in the intergovernmental review process, and therefore do not have a SPOC. If you are located within one of these States, you may still send application materials directly to a Federal awarding agency.

#### Arkansas

Tracy L. Copeland Manager, State Clearinghouse Office of Intergovernmental Services Department of Finance and Administration 1515 W. 7th St., Room 412 Little Rock, Arkansas 72203 Telephone: (501) 682–1074 Fax: (501) 682–5206 tlcopeland@dfa.state.ar.us

#### California

Grants Coordination State Clearinghouse Office of Planning and Research P.O. Box 3044, Room 222 Sacramento, California 95812–3044 Telephone: (916) 445–0613 Fax: (916) 323–3018 state.clearinghouse@opr.ca.gov

#### Delaware

Charles H. Hopkins Executive Department Office of the Budget 540 S. Dupont Highway, 3rd Floor Dover, Delaware 19901 Telephone: (302) 739–3323 Fax: (302) 739–5661 chopkins@state.de.us

#### District of Columbia

Ron Seldon Office of Grants Management and Development 717 14th Street, NW, Suite 1200 Washington, DC 20005 Telephone: (202) 727–1705 Fax: (202) 727–1617 ogmd-ogmd@dcgov.org

#### Florida

Florida State Clearinghouse Department of Community Affairs 2555 Shumard Oak Blvd. Tallahassee, Florida 32399–2100 Telephone: (850) 922–5438 (850) 414–5495 (direct) Fax: (850) 414–0479

#### Georgia

Georgia State Clearinghouse 270 Washington Street, SW Atlanta, Georgia 30334 Telephone: (404) 656–3855 Fax: (404) 656–7901 gach@mail.opb.state.ga.us

#### Illino

Virginia Bova Department of Commerce and Community Affairs James R. Thompson Center 100 West Randolph, Suite 3–400 Chicago, Illinois 60601 Telephone: (312) 814–6028

#### Iow

Fax (312) 814-8485

vbova@commerce.state.il.us

Steven R. McCann Division of Community and Rural Development Iowa Department of Economic Development 200 East Grand Avenue Des Moines, Iowa 50309 Telephone: (515) 242–4719 Fax: (515) 242–4809 steve.mccann@ided.state.ia.us

#### Kentucky Ron Cook

Department for Local Government 1024 Capital Center Drive, Suite 340 Frankfort, Kentucky 40601 Telephone: (502) 573–2382 Fax: (502) 573–2512 ron.cook@mail.state.ky.us

#### Maine

Joyce Benson State Planning Office 184 State Street 38 State House Station Augusta, Maine 04333 Telephone: (207) 287–3261 (207) 287–1461 (direct) Fax: (207) 287–6489 joyce.benson@state.me.us

#### Maryland

Linda Janey Manager, Clearinghouse and Plan Review Unit

Maryland Office of Planning 301 West Preston Street—Room 1104 Baltimore, Maryland 21201–2305 Telephone: (410) 767–4490 Fax: (410) 767–4480 linda@mail.op.state.md.us

#### Michigan

Richard Pfaff Southeast Michigan Council of Governments 535 Griswold, Suite 300 Detroit, Michigan 48226 Telephone: (313) 961–4266 Fax: (313) 961–4869 pfaff@semcog.org

Mississippi Cathy Mallette Clearinghouse Officer Department of Finance and Administration 1301 Woolfolk Building, Suite E 501 North West Street Jackson, Mississippi 39201 Telephone: (601) 359-6762 Fax: (601) 359-6758

Missouri Lois Pohl

Federal Assistance Clearinghouse Office of Administration P.O. Box 809 Tefferson Building, Room 915 Jefferson City, Missouri 65102 Telephone: (573) 751–4834 Fax: (573) 522-4395 pohll @mail.oa.state.mo.us

Nevada

Heather Elliott Department of Administration State Clearinghouse 209 E. Musser Street, Room 200 Carson City, Nevada 89701 Telephone: (775) 684-0209 Fax: (775) 684-0260 helliott@govmail.state.nv.us

New Hampshire

Jeffrev H. Taylor Director

New Hampshire Office of State Planning Attn: Intergovernmental Review Process Mike Blake

21/2 Beacon Street

Concord, New Hampshire 03301 Telephone: (603) 271-1728 Fax: (603) 271-1728 jtaylor@osp.state.nh.us

New Mexico

Ken Hughes Local Government Division Room 201 Bataan Memorial Building Santa Fe, New Mexico 87503 Telephone: (505) 827-4370

Fax: (505) 827-4948 khughes@dfa.state.nm.us

North Carolina

Jeanette Furney Department of Administration 1302 Mail Service Center Raleigh, North Carolina 27699–1302 Telephone: (919) 807-2323 Fax: (919) 733-9571 jeanette.furney@ncmail.net

North Dakota

Jim Boyd Division of Community Services 600 East Boulevard Ave, Dept 105 Bismarck, North Dakota 58505-0170 Telephone: (701) 328-2094 Fax: (701) 328-2308

jboyd@state.nd.us

Rhode Island

Kevin Nelson Department of Administration Statewide Planning Program

One Capitol Hill

Providence, Rhode Island 02908-5870 Telephone: (401) 222-2093

Fax: (401) 222-2083 knelson@doa.state.ri.us South Carolina

Omeagia Burgess Budget and Control Board Office of State Budget 1122 Ladies Street, 12th Floor Columbia, South Carolina 29201 Telephone: (803) 734-0494 Fax: (803) 734-0645 aburgess@budget.state.sc.us

Texas

Denise S. Francis Director, State Grants Team Governor's Office of Budget and Planning P.O. Box 12428 Austin, Texas 78711 Telephone: (512) 305-9415 Fax: (512) 936-2681 dfrancis@governor.state.tx.us

Utah

Carolyn Wright Utah State Clearinghouse Governor's Office of Planning and Budget State Capitol, Room 114 Salt Lake City, Utah 84114 Telephone: (801) 538-1535 Fax: (801) 538-1547 cwright@gov.state.ut.us

West Virginia

Fred Cutlip, Director Community Development Division West Virginia Development Office Building #6, Room 553 Charleston, West Virginia 25305 Telephone: (304) 558-4010 Fax: (304) 558-3248 fcutlip@wvdo.org

Wisconsin

Jeff Smith

Section Chief, Federal/State Relations Wisconsin Department of Administration 101 East Wilson Street-6th Floor P.O. Box 7868

Madison, Wisconsin 53707 Telephone: (608) 266-0267 Fax: (608) 267-6931 jeffrey.smith@doa.state.wi.us

American Samoa

Pat M. Galea'i

Federal Grants/Programs Coordinator Office of Federal Programs Office of the Governor/Department of

Commerce

American Samoa Government Pago Pago, American Samoa 96799 Telephone: (684) 633-5155 Fax: (684) 633-4195 pmgaleai@samoatelco.com

Guam

Director Bureau of Budget and Management Research Office of the Governor P.O. Box 2950 Agana, Guam 96910 Telephone: 011-671-472-2285 Fax: 011-472-2825 jer@ns.gov.gu

Puerto Rico

Jose Caballero/Mayra Silva Puerto Rico Planning Board

Federal Proposals Review Office Minillas Government Center P.O. Box 41119 San Juan, Puerto Rico 00940-1119

Telephone: (787) 723-6190 Fax: (787) 722-6783

North Mariana Islands

Ms. Jacoba T. Seman Federal Programs Coordinator Office of Management and Budget Office of the Governor Saipan, MP 96950 Telephone: (670) 664-2289 Fax: (670) 664-2272 omb.jseman@saipan.com

Virgin Islands

Ira Mills

Director, Office of Management and Budget #41 Norre Gade Emancipation Garden

Station, Second Floor Saint Thomas, Virgin Islands 00802 Telephone: (340) 774-0750 Fax: (340) 776-0069 Irmills@usvi.org

Changes to this list can be made only after OMB is notified by a State's officially designated representative. E-mail messages can be sent to grants@omb.eop.gov. If you prefer, you may send correspondence to the following postal address: Attn: Grants Management, Office of Management and Budget, New Executive Office Building, Suite 6025, 725 17th Street, NW, Washington, DC.

#### Appendix B—Competency Goals and **Indicators for Head Start Staff Working** With Families

The "Competency Goals and Indicators for Head Start Staff Working with Families' described on the following pages are intended to define competencies and skills for entry-level staff who are working directly with families under ongoing supervision in furtherance of their professional development. Family Workers should be able to demonstrate their ability to provide services consistent with the requirements of the Head Start Program Performance Standards.

Today's workers are expected to exhibit a new level of professionalism to effectively support today's families. Increasingly, new organizational structures and innovative service models within Head Start require workers to:

- Develop respectful relationships with families which evolve into an individualized family partnering process which addresses the parent's role in supporting child development goals, health and disabilities goals, as well as traditional social services, family development, and parent involvement goals.
- · Work in partnership with families and other community providers to develop family partnership agreements and to integrate this process into family plans when appropriate.
- Support families in their efforts to obtain employment and move towards selfsufficiency.
- Provide a new level of service in the area of family literacy, reflective of the intent of the current Head Start Program Performance Standards.

Appendix B as follows:

- Reflects the Head Start Program Performance Standards
- Reflects the latest thinking in the family support field including strength-based, family centered principles, and

• Includes new areas of competency in response to the changing role of family support staff

Indicators are listed for each area of competency. These Indicators provide a mechanism to measure individuals seeking demonstrable competency in each of the competency goal areas.

BILLING CODE 4184-01-P

#### COMPETENCY GOALS AND INDICATORS FOR HEAD START STAFF WORKING WITH FAMILIES

<u>Competency Goal #1:</u> Establish mutually respectful partnerships with families to enhance the quality of their lives and their communities.

<u>Indicators:</u> Head Start staff working with families should demonstrate their ability to do the following:

- Conduct outreach, recruitment, and enrollment;
- Provide orientation on philosophy of program and services provided;
- Establish and maintain ongoing partnerships based on trust with families;
- Communicate effectively using appropriate verbal and nonverbal messages and reflective listening skills;
- Implement strategies including home visits to learn about families and the changing community.

### **Competency Goal #2:** Support families' efforts to reach their goals.

<u>Indicators:</u> Head Start staff working with families should demonstrate their ability to do the following:

- Develop strengths-based assessments with families that describe their goals, strengths, resources and support networks, as well as necessary services and supports;
- Develop, in partnership with the family, an individualized family plan;
- Facilitate families' problem-solving and teach problem-solving skills;
- Coach, consult, educate, and utilize counseling skills, where appropriate;
- Advocate for the family and support them in advocating for themselves;
- Follow-up with the family on the progress toward meeting their goals and any needed revisions to the plan;
- Assist with transitions to other programs, communities and schools.

<u>Competency Goal #3:</u> Offer parents opportunities to be involved in group activities,

including policy groups and educational activities based on

interest and need.

<u>Indicators:</u> Head Start staff working with families should demonstrate their ability to do the following:

- Identify common interests and needs of parents in order to plan appropriate activities;
- Work with parent groups on group formation, group processing, and leadership
- Provide and/or coordinate training and educational opportunities for parents; and
- Engage parents in volunteering, community service and other ways of contributing to program activities and services.

<u>Competency Goal #4:</u> Provide opportunities for children and families to participate in family literacy services.

**Indicators:** Head Start staff working with families should demonstrate their ability to do the following:

- Work with other program staff to support interactive literacy activities between parents and their children;
- Provide training for parents in how to be the primary teacher for their children and full partners in the education of their children:
- Assist parents as adult learners to recognize and address their own literacy goals; and
- Link and support parents in engaging in literacy training that contributes to self-sufficiency.

Competency Goal #5: Coordinate and integrate Head Start services in order to enhance effectiveness.

<u>Indicators:</u> Head Start staff working with families should demonstrate their ability to do the following:

- Serve as a productive team member on an interdisciplinary team of professionals;
- Participate in and facilitate case conferences to promote service integration;
- Apply knowledge of health, mental health, disabilities, and child development in order to ensure holistic service delivery;
- Promote and support parent involvement and leadership throughout the program.

### **Competency Goal #6:** Support families in accessing other community resources.

<u>Indicators:</u> Head Start staff working with families should demonstrate their ability to do the following:

- Research and keep up-to-date on program and community resources;
- Analyze match of community resources to family needs and identify unmet family needs.
- Refer families to community resources and follow-up on the effectiveness of referrals; and
- Promote community partnerships that will improve supports to families.

#### Competency Goal #7: Assist families in crisis.

<u>Indicators:</u> Head Start staff working with families should demonstrate their ability to do the following:

- Listen to families and assess the crisis situation;
- Take active steps to ensure the safety of all involved;
- Decide when to intervene and when to refer a family;
- Identify (with the family) options, resources, and consequences to address the crisis; and
- Support families in making decisions and taking active steps to resolve current crises and be prepared to address future crises;

Competency Goal #8: Respect and respond competently to the culture, traditions, lifestyle, language, and values of each family and community.

Indicators: Head Start staff working with families should demonstrate their ability to do the following:

- Be knowledgeable about and sensitive to each family's values, beliefs, traditions, cultural influences, makeup, and circumstances;
- Work with families representing different cultures using a culturally competent and flexible approach.
- Identify and reflect on personal values, experiences and biases that facilitate and present barriers in working with certain groups of people.

# Competency Goal #9 Contribute to effective program practices and maintain a commitment to professionalism.

<u>Indicators:</u> Head Start staff working with families should demonstrate their ability to do the following:

- Perform record-keeping and internal and external reporting tasks in a timely and objective fashion;
- Effectively utilize supervisory professional development and technical assistance resources to improve competence;
- Contribute to and participate in strategic planning, program self-assessment and other efforts to improve program services and agency responsiveness to families;
- Make decisions and act based on family support principles, theories, practices, and code of ethics;
- Articulate an awareness of self, values, and ethics as they impact on work with families; and
- Maintain professional boundaries and confidentiality.

Appendix C—Head Start Quality Improvement Centers

Region/States	Agency	Project Director
<b>Region I</b> Connecticut Maine	Education Development Center, Inc. 55 Chapel Street Newton, MA 02458-1060	Pat Fahey 617-969-7100 x2375 617-969-3440 Fax E-mail[pfahey@edc.org]
Massachusetts New Hampshire Rhode Island Vermont	Disabilities Services: Education Development Center, Inc. 55 Chapel Street Newton, MA 02458-1060	Philip Printz  800-225-4276 x2349  617-618-2349  617-969-3440 Fax E-mail[pprintz@edc.org]
Region II (a) New Jersey	(a) New York University Head Start Quality Improvement Center 726 Broadway, 5 <sup>th</sup> Floor New York, NY 10003	Patricia M. Hall 212-998-5550 212-995-3458 Fax E-mail[pmh5@is2.nyu.edu]
New York  (b)  Puerto Rico/ Virgin Islands	(b) Development Associates, Inc. Puerto Rico/Virgin Islands P.O. Box 3968 Guaynabo, PR 00970-3968	Gloria de-Llovio Dominguez 787-281-0125/0100/0110 787-281-0120 Fax E-mail[dahstasc@tld.net]
virgii Istalius	Disabilities Services:  New York University School of Education 726 Broadway, 5° Floor New York, NY 10003	Barbara Schwartz 212-998-5528 212-995-4562 Fax E-mail[b.schwartz@nyu.edu]
	Satellite Office  New York University QIC-D  130 Winston Churchill Avenue San Juan, PR 00926	Linna Irizarry-Mayoral 787-751-8880 787-751-8810 Fax E-mail[cynuqicd@tld.net]
<b>Region III</b> Delaware  District of Columbia  Maryland  Pennsylvania	University of Maryland University College 3501 University Boulevard East Adelphi, MD 20783	Madhavi Parikh 301-446-1040 800-688-1675 301-446-1060 Fax E-mail[parim@hsrtc.umuc.edu]
Virginia West Virginia	Child Development Resources P.O. Box 280 Norge, VA 23127-0280  Satellite Office	Sheri Osborne 757-566-3300 757-566-8977 Fax E-mail [sherio@cdr.org]
	West Virginia DSQIC Office 2A Hyre Avenue Petersburg, WV 26847	Carol Thomas 304-257-4853 888-869-1359 304-257-4829 Fax E-mail[carolm@hardynet.com]

Region/States	Agency	Project Director
Region III (Cont.)  Delaware  District of Columbia  Maryland	Children's Hospital and the UCLID Center University of Pittsburgh 3705 Fifth Avenue at Desoto Street Pittsburgh, PA 15123-2583	Stephen J. Bagnato 412-692-6520 412-692-8729 Fax E-mail[stephen_bagnato@poplar.chp.edu]
Pennsylvania Virginia West Virginia	Subcontract Office Georgetown University Child Development Center 3307 M Street, SW, 4th Floor Washington, DC 20007-3935	Kris Hansen 202-687-5107 202-687-8899 Fax E-mail[hansenk@gunet.georgetown.edu]
Region IV  (a)  Kentucky North Carolina South Carolina Tennessee	(a) Western Kentucky University Training and Technical Assistance Services College of Education Room 344, Tate C. Page Hall 1 Big Red Way Bowling Green, KY 42101-3576	Colleen B. Mendel 270-745-4041 270-745-3340 Fax E-mail[colleen.mendel@wku.edu]
(b) Alabama Florida Georgia Mississippi	(b) Western Kentucky University Training and Technical Assistance Services College of Education Room 344, Tate C. Page Hall 1 Big Red Way Bowling Green, KY 42101-3576	Marce Verzaro-O'Brien 305-289-2034 800-882-7482 305-289-0337 Fax E-mail[mvobrien@aol.com]
	Disabilities Services: Chapel Hill Training-Outreach Project 800 Eastowne Drive, Suite 105 Chapel Hill, NC 27514  Satellite Offices	Brenda Bowen 919-490-5577 x224 800-473-1727 919-490-4905 Fax E-mail[bbowen@intrex.net]
	Florida  Pinellas County Head Start  66988 <sup>th</sup> Avenue North, Suite D  Pinellas Park, FL 33781	Pam Kautz 727-547-5900 727-547-5909 Fax E-mail[pamkautz@aol.com]
	Georgia 1806 Calibre Woods Drive Atlanta, GA 30329	Lisa Goldman 404-325-0903 404-329-5488 Fax E-mail[lmgoldman@mindspring.com]
	Mississippi 4791 McWillie Drive, Suite 103 Jackson, MS 39206	Valerie Campbell 601-362-9154 601-362-9487 Fax E-mail[campbell@misnet.com]

Region/States	Agency	Project Director
Region V	(a) Cooperative Educational Service Agency #5	Linda Young
, .	P.O. Box 564	608-742-8814 x309
(a)	626 East Slifer Street	800-862-3275 x309
	Portage, WI 53901	608-742-2384 Fax
Michigan	6.4 W. 060	E-mail[youngl@cesa5.k12.wi.us]
Minnesota	Satellite Offices	
Wisconsin	Remidii Caste University	Connie Grant
(b)	Bemidji State University 325 Education/Art Building (EA325)	218-755-3782
(6)	1500 Birchmont Drive NE	218-755-3787 Fax
Illinois	Bemidji, MN 56601-2699	E-mail[cegrant@vax1.bemidji.msus.edu]
Indiana	Delings, AM 30001-2099	L man(cegrant@vax1.bemaji.msus.edu)
Ohio	BHK Head Start	Sally Hruska
	HC2 Box 989	906-337-5043
	Allouez, MI 49805	906-337-5043 Fax
		E-mail[shruska@up.net]
	Michigan State University	Brooke Foulds
	Kellogg Center Room 22	517-353-5194
	East Lansing, MI 48824-1036	517-355-5220 Fax
		E-mail[fouldsdo@pilot.msu.edu]
	University of Minnesota	Jean Marchessault
	370C Peik Hall	612-624-9509
	159 Pillsbury Drive	612-625-2097 Fax
	Minneapolis, MN 55455-0223	E-mail[march020@tc.umn.edu]
	UW-Waisman Center	Linda Leonhart
	122 E. Olin Avenue	608-265-9423
	Suite 110	608-265-9421 Fax
	Madison, WI 53713	E-mail[leonhart@waisman.wisc.edu]
		Dennis Sykes
	(b) The Ohio State University	614-447-0844 x133
	Center for Special Needs Populations	800-447-1424
	Suite 440	614-447-9043 Fax
	700 Ackerman Road	E-mail[sykes.3@osu.edu]
	Columbus, OH 43202-1559	z manijajnesia (e osuletau)
	Satellite Offices	
	OHSAI	
	505 Windsor Park Drive	Barbara Haxton
	Dayton, OH 45459	937-435-1113
		937-435-5411 Fax
		E-mail[haxton@erinet.com]

Region/States	Agency	Project Director
inegrony states		
Region V (Cont.)	OSU College of Education	David Fernie
	ARPS Hall, Room 333	614-292-8023
(a)	1945 North High Street	614-292-7695 Fax
· ·	Columbus, OH 43202	E-mail[fernie.1@osu.edu]
Michigan		
Minnesota	University of Cincinnati	Larry Johnson
Wisconsin	Evaluations Service Center	513-556-2322
	P.O. Box 210002	513-556-9311 Fax
(b)	One Edwards Center	E-mail[Lawrence.Johnson@uc.edu]
	51 W. Corry Street	
Illinois	Cincinnati, 0H 45221-0105	
Indiana		
0hio		
	Disabilities Services:	
	University of Illinois	Teresa C. Bennett
	Department of Special Education	217-333-3876
	139 Children's Research Center, 51 Gerty Drive	217-244-6191 Fax
	Champaign, IL 61820	E-mail[t-benne@uiuc.edu]
	Subcontract Offices	
	Chicago	
	Chicago Great Lakes QIC-D Office	Bea Nichols
11 11	P.O. Box 436945	773-429-9723
	Chicago, IL 60643-6945	773-429-9724 Fax
	ceago, 22 00043 0343	
	Ohio	
	Ohio Head Start Association, Inc.	Barbara Haxton
	505 Windsor Park Drive	937-435-1113
	Dayton, OH 45459	937-435-5411 Fax
	, , , , , , , , , , , , , , , , , , , ,	E-mail[haxton@erinet.com]
	Wisconsin	
	CESA 5	Nola Larson
	626 E. Slifer Street	608-742-8814
	P.O. Box 564	608-742-2384 x233
	Portage, WI 53901	E-mail[larsonn@cesa5.k12.wi.us]
D - 1 17		
Region VI	(a) Basic Health Management	Linda Reasoner
	The Lafayette Building	501-370-9155
(a)	523 South Louisiana, Suite 303	800-270-2872
A al	Little Rock, AR 72201	501-370-9158 Fax
Arkansas		E-mail[lreasoner@bhmqic.com]
Louisiana Oklahoma		James Mikelinit
UKIANUMA	(b) Texas Tech University	James Mitchell
(6)	Institute for Child and Family Studies	806-742-3296
(b)	College of Human Sciences, Room 167	800-527-2802
New Mexico	Box 41162	806-742-0508 Fax
Texas	Lubbock, TX 79409-1162	E-mail[jmitchellttu@worldnet.att.net]
15/83		

Region/States	Agency	Project Director
Region VI (Cont.)  (a)  Arkansas Louisiana Oklahoma  (b)  New Mexico Texas	Disabilities Services: University of Arkansas UAP 501 Woodlane, Suite 210 Little Rock, AR 72201	Karan Burnette 800-831-4827 501-682-9900 501-682-9901 Fax E-mail[BurnetteKaranB@exchange.uams.edu]
Region VII  Iowa  Kansas  Missoun  Nebraska	Community Development Institute 5616 Raytown Road Raytown, MO 64133  Disabilities Services: University of Kansas Medical Center KUMC CDU G001 3901 Rainbow Boulevard Kansas City, KS 66160-7339	Donna McDaniel 816-356-5373 816-356-2818 Fax E-mail[Dmcdaniel@CDI7.com] Barbara Lawrence 913-588-5960 913-588-5920 Fax E-mail[blawren1@kumc.edu]
Region VIII  Colorado  Montana North Dakota South Dakota Utah Wyoming	Community Development Institute 9745 E. Hampden Avenue, Suite 310 Denver, CO 80231  Disabilities Services: Utah State University/HSRC Room 13 6582 Old Main Hill Logan, UT 84322-6582	Deborah Hinrichs 303-369-5959 x19 800-488-CDI8 (2348) 303-369-5801 Fax E-mail[Debhinrich@aol.com] or [CDI8QIC@aol.com]  Don Barringer 435-797-0988 888-523-5127 435-797-0983 Fax
	Subcontract Offices  PEAK Parent Center, Inc. 6055 Lehman Drive, Suite 101 Colorado Springs, CO 80918  Parents Let's Unite for Kids (PLUK) 516 N. 32™ Street Billings, MT 59101-0298	E-mail[nordon@cc.usu.edu]  Shirley Swope 719-531-9400 719-531-9452 Fax E-mail[parentadvisor@peakparent.org]  Rebecca Johns 406-837-2029 888-288-2029 406-255-0253 Fax E-mail[rjohns@digisys.net]

Region/States	Agency	Project Director
Region VIII (Cont.)	Pathfinder Family Center (Pathfinder PTI)	Kathryn Erikson
	Arrowhead Mall	701-837-7500
Colorado	1600 2 <sup>™</sup> Avenue SW	701-837-7501 TDD
Montana	Suite 19	701-837-7548 Fax
North Dakota	Minot, ND 58701-3459	E-mail[ndpath01@minot.ndak.net]
South Dakota		
Utah		
Wyoming	South Dakota Parent Connection	Bev Peterson
	3701 West 49th Street	605-361-3171
	Suite 200B	605-361-2928 Fax
	Sioux Falls, SD 57106	E-mail[bpeterson@dakota.net]
	Utah Parent Center	Jennie Gibson
	2290 East 4500 South, Suite 110	801-763-0722
	Salt Lake City, UT 84117	801-763-0422 Fax
		Terri Dawson
	Parent Information Center	307-684-2277
	5 North Lobban	307-684-2277 307-684-5314 Fax
	Buffalo, WY 82834	E-mail[tdawson@vcn.com]
		E-man(tuawson@vcn.com)
Region IX	Development Associates, Inc.	Kelly M. Crowell, Jr.
megren in	1475 North Broadway, Suite 200	925-935-9711
Arizona	Walnut Creek, CA 94596	800-666-9711
California	water creen, as 3 issue	925-935-0413 Fax
Guam		E-mail[kcrowell@devassoc.com]
Hawaii	Satellite Offices	Web[www.devassoc.com]
Nevada		
Outer Pacific Area	Arizona	
	Quality Improvement Center at the	Norbert C. Corbeille, Jr.
	Arizona State Head Start Association	480-829-6335
	3910 S. Rural Road, Suite M	480-829-8768 Fax
	Tempe, AZ 85282	E-mail[ncorbeil@devassoc.com]
	California	
	Development Associates, Inc.	Jacqueline Davis
	Quality Improvement Center at the	800-666-9711
	Research Center	626-792-9720
	Pacific Oaks College	626-397-1304 Fax
	65 S. Grand Avenue	E-mail[jdavis@devassoc.com]
	Pasadena, CA 91103-3592	-

Region/States	Agency	Project Director
	-	
Region IX (Cont.)	Development Associates, Inc.	Stephen Romano
	Quality Improvement Center at the	800-666-9711
Arizona	Neighborhood House Association	619-715-2642 x158
California	5660 Copley Drive	619-715-5830 Fax
Guam	San Diego, CA 92111	E-mail[sromano@devassoc.com]
Hawaii		
Nevada		
Outer Pacific Area	Development Associates, Inc.	
	Quality Improvement Center	
	4812 N. Fourth Street	
	Fresno, CA 93726	
	Development Associates, Inc.	
	1475 North Broadway, Suite 200	
	Walnut Creek, CA 94596	
	Disabilities Services:	
	California Institute of Human Services	Mary Anne Doan
	Sonoma State University	707-664-2957
	1801 East Cotati Avenue	800-625-7648
	Rohnert Park, CA 94928-3609	707-664-2418 Fax
	Ronnert Park, CA 94928-3009	E-mail[maryanne.doan@sonoma.edu]
	Southern Office	
	Ventura County Superintendent of	Mary Ann Walker
	Schools Office	805-383-9301
		800-625-7649
	5189 Verdugo Way	805-383-9304 Fax
	Camarillo, CA 93012	E-mail[mawalker@vcss.k12.ca.us]
		Web[http://www.sonoma.edu/cihs/hsds/]
Region X	Farly Childhood Training Contact	Carillon J. Olmsted
region A	Early Childhood Training Center	l 1
Alaska	Portland State University P.O. Box 1491	503-725-4815
		503-725-4838 Fax
Idaho Oregon	1633 SW Park Avenue	E-mail[olmstedc@pdx.edu]
Washington	Portland, OR 97207	Web[http://extended.pdx.edu/ectc.htm]
washington	<u>Sub-Grant</u>	
	Alaska Satellite Support Center	Sally Mead
	Prevention Associates	907-272-6925
		1
	101 East 9th Avenue, Suite 7A	907-272-6946 Fax
L	Anchorage, AK 99501	E-mail[smead@alaska.net]

July 2000

Region/States	Agency	Project Director
<b>Region X (Cont.)</b> Alaska Idaho Oregon Washington	Disabilities Services: Early Childhood Training Center Portland State University P.O. Box 1491 1633 SW Park Avenue Portland, OR 97207	Carillon J. Olmsted 503-725-4815 503-725-4838 Fax E-mail[olmstedc@pdx.edu] Web[http://extended.pdx.edu/ectc.htm]
Region XI	University of Oklahoma American Indian Institute 555 Constitution Street Suite 237 Norman, OK 73072-7820  Disabilities Services: Three Feathers Associates P.O. Box 5508 Norman, OK 73070	Pattie Howell 405-325-4129 405-325-7319 Fax E-mail[phowell@ou.edu]  Antonia Dobrec 405-360-2919 405-360-3069 Fax E-mail[toni@threefeathersassoc.com]
Region XII	Academy for Educational Development 1825 Connecticut Avenue, NW Washington, DC 20009-5721  Disabilities Services: Academy for Educational Development 1825 Connecticut Avenue, NW Washington, DC 20009-5721	Leilani Pennel 800-864-0465 202-884-8729 202-884-8732 Fax E-mail[Lpennel@aed.org]  Julie Jones 800-864-0465 202-884-8728 202-884-8732 Fax E-mail[jjones@aed.org]

[FR Doc. 01–10901 Filed 5–2–01; 8:45 am] BILLING CODE 4184–01–C